

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Anghel, Brindusa; Balart, Pau

Article

Non-cognitive skills and individual earnings: New evidence from PIAAC

SERIEs - Journal of the Spanish Economic Association

Provided in Cooperation with: Spanish Economic Association

Suggested Citation: Anghel, Brindusa; Balart, Pau (2017) : Non-cognitive skills and individual earnings: New evidence from PIAAC, SERIEs - Journal of the Spanish Economic Association, ISSN 1869-4195, Springer, Heidelberg, Vol. 8, Iss. 4, pp. 417-473, https://doi.org/10.1007/s13209-017-0165-x

This Version is available at: https://hdl.handle.net/10419/195265

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



ORIGINAL ARTICLE



Non-cognitive skills and individual earnings: new evidence from PIAAC

Brindusa Anghel¹ · Pau Balart²

Received: 27 July 2017 / Accepted: 27 October 2017 / Published online: 14 November 2017 © The Author(s) 2017. This article is an open access publication

Abstract We use the Programme for the International Assessment of Adult Competencies (PIAAC) data to construct several measures of non-cognitive skills and to analyze the relationship between non-cognitive skills and earnings. We construct measures for non-cognitive skills based on previous research in the field, such as selfreports and non-response in the background questionnaire. We also take advantage of the computer based nature of the assessment to explore the effects of other constructs such as skipped questions in the test, average time per question, average time per correct answer and the number of keyboard and mouse movements before answering a question. We find that, even after controlling for PIAAC scores, previously proposed proxies of non-cognitive skills (self-reports and survey non-response) are significantly associated with individual earnings. We also use these measures of non-cognitive skills to propose a correction of the PIAAC scores as a measure of cognitive skills in the earnings equation. By doing so, we observe that the relationship between earnings and PIAAC scores might be partially mediated by non-cognitive skills. However, our results also show that PIAAC test scores remain highly significant and as the main determinant of earnings after the correction.

Keywords Non-cognitive skills · Earnings · PIAAC data

Brindusa Anghel brindusa.anghel@bde.es

> Pau Balart pau.balart@uib.cat

- ¹ Banco de España, Madrid, Spain
- ² Universitat de les Illes Baleares, Palma, Spain

Electronic supplementary material The online version of this article (https://doi.org/10.1007/s13209-017-0165-x) contains supplementary material, which is available to authorized users.

JEL Classification I21 · I26

1 Introduction

Despite the important role that human capital occupies in economic thinking, its nonobservable nature has historically limited its measurement. Since the original works by Becker (1962) and Mincer (1970, 1974), years of schooling has remained as the predominant measure of human capital. It was with the improvement in testing techniques and after the works by Hanushek, Woessmann and their coauthors (see Hanushek and Woessmann 2008, 2011, among others) that the use of international standardized achievement tests started to be seen as a more adequate measure of human capital. As argued by Hanushek and Woessmann (2008) achievement tests provide a better measure of human capital than years of schooling because (1) they better capture skills acquired out of the classroom and (2) because there is substantial cross-country variation in skills acquired at each level of education. Nowadays, there is a growing attention towards the distinction between two different types of skills that are embedded in human capital, namely cognitive and non-cognitive skills. A better understanding of the importance of human capital requires the construction of measures that distinguish between these two types of skills.

We understand by cognitive skills individuals' knowledge on basic competences, such as numeracy and literacy. Non-cognitive skills (also known as soft skills or socioemotional skills) are individual patterns of behaviors, attitudes and personality that are not directly related to individuals' knowledge. Recent research has found that this distinction is important when studying the relationship between human capital and economic outputs both at an individual (Duckworth et al. 2011; Kautz et al. 2014; Heckman and Rubinstein 2001; Heckman et al. 2006, 2010) as well as at an aggregate level (Brunello and Schlotter 2011). At the same time, as noted by Cunha et al. (2010), the type of interventions required to foster cognitive and non-cognitive skills are notably different which intensifies the necessity of a better understanding of the role of each type of skills for policy recommendations.¹

The main challenge for the study of non-cognitive skills is often the availability of proper measures. This is not the case for cognitive skills. The evaluation of cognitive skills notably improved during the last decades of the twentieth century, thanks to the advance in testing techniques. This fostered the emergence of standardized achievement tests allowing for international comparisons of cognitive skills. In contrast, non-cognitive skills and personality traits are less present in internationally comparable datasets. First, because the measurement of non-cognitive skills has been traditionally based on self-reports which raises several methodological concerns due to misreports, difficulties in interpersonal comparisons or the socially desirable response bias (see King et al. 2004; Paulhus 1984; Soest et al. 2011). Second, because attention to these types of skills is relatively new and started to grow at the beginning of the twentyfirst century with the works by Heckman and his coauthors. However, recent research has provided alternative measures of non-cognitive skills. First, Borghans et al. (2011),

¹ See García (2016) for an overview of the evidence on policy interventions to foster non-cognitive skills.

Duckworth et al. (2011) and Segal (2012) show that scores on low-stake achievement tests are not only driven by high cognitive skills but also by high non-cognitive skills. This observation provides an opportunity to have more detailed measures of human capital that specifically distinguish between cognitive and non-cognitive skills. Second, there is an increasing research towards the possibility of using testing and survey behavior as a measure of non-cognitive skills. Hedengren and Stratmann (2012) and Hitt et al. (2016) proposed and validated survey non-response as a measure of non-cognitive traits of the big five taxonomy, predict patterns of survey non-response.² Zamarro et al. (2016a, b, 2017) further develop and review non-response as a measure of non-cognitive skills.

The above findings raise several questions regarding the measurement of human capital. The first one is, how should one interpret scores of achievement tests? According to the conclusions of Borghans et al. (2011), Duckworth et al. (2011) and Segal (2012), scores on low-stakes achievement tests should not be interpreted as a measure of cognitive skills but as a broader notion of human capital that also includes non-cognitive skills. The second question concerns the possibility of obtaining separate measures for each type of skill. For instance, Borghans and Schils (2012) proposed a methodology to separate PISA scores into a cognitive and non-cognitive component. Third, and based on the findings of Hedengren and Stratmann (2012), Hitt et al. (2016), Cheng et al. (2016) and Zamarro et al. (2016a, b, 2017) one may ask about the possibility of using some already available and internationally comparable assessments to create and study non-cognitive skills.

In the present paper we use the PIAAC database to construct several measures of non-cognitive skills and include them in the Mincerian equation. We construct different types of measures for non-cognitive skills. On the one hand, we use the answers in three items of the PIAAC background questionnaire to obtain self-reported measures of non-cognitive skills. On the other hand, we follow Hedengren and Stratmann (2012) and Hitt et al. (2016) to construct measures based on survey non-response in the background questionnaire. Finally, we use some other measures based on the testing behavior rather than on the background questionnaire. In particular, we take advantage of the computer based nature of the PIAAC test, to construct measures that might be related to non-cognitive skills, such as time spent on each item in the test, skipped questions or the number of actions done with the mouse before answering an item. To the best of our knowledge, only skipped test items has been used before as a proxy for non-cognitive skills by Hernández and Hershaff (2014).

We use the proposed measures to analyze the association between non-cognitive skills and earnings in a Mincerian equation. Our results show that measures of noncognitive skills proposed in previous literature (self-reports and non-response in a questionnaire) are significantly associated to earnings in all our specifications (with the exception of self-reports on cultural engagement). When looking at test-based measures, we find that skipped test items and not-attempted items are negatively and

² The association between non-response patterns and non-cognitive skills was already suggested by Groves et al. (2011) and Boe et al. (2002). Using PISA data, Balart (2017) shows that parental non-response in a questionnaire is negatively associated with their children's tests scores.

significantly associated to earnings in all specifications. However, these two measures are the ones that exhibit higher correlation with test scores so they might be capturing cognitive skills. Average time per correct answer is also statistically correlated to earnings. Additionally, we perform various robustness checks to test the validity of our measures of non-cognitive skills.

We also pay attention to the relationship between cognitive skills and earnings and attempt to provide an estimate for the effects of cognitive skills in the Mincerian equation after correcting for the effect of non-cognitive skills. In a recent article, Hanushek et al. (2015) use data from the PIAAC database to study the relationship between human capital and individual earnings. They show that PIAAC scores have explanatory power on earnings above and beyond years of schooling (the traditional measure of human capital). However, as emphasized in the psychology field (Wechsler 1940) and in the economics literature (Duckworth et al. 2011; Segal 2012) performance on achievement tests depends not only on cognitive but also on non-cognitive skills.³ Therefore, it is not clear until what extent the association between PIAAC scores and earnings found by Hanushek et al. (2015) is driven by cognitive or non-cognitive factors. In other words, interpreting the results of Hanushek et al. (2015) in terms of cognitive skills may be problematic. We propose a correction of the PIAAC test scores using our proxies for non-cognitive skills. To do so, first we regress PIAAC test scores on non-cognitive skills and, in a second stage, we use the residuals of the previous regression as a corrected measure of cognitive skills. This correction is based on the finding by Cunha and Heckman (2008) and Cunha et al. (2010) that non-cognitive skills foster the development of cognitive skills, while the opposite is not true. Thus, the proposed measure only takes into account variation in cognitive skills that is orthogonal to our measures of non-cognitive skills. Despite reducing the size of the estimates of cognitive skills, cognitive skills remain highly significant and the main determinant of individual earnings after this (over) correction of PIAAC test scores.

As cautioned by Hanushek et al. (2015) an important drawback of studies based on cross-sectional international data such as PIAAC relies on the impossibility of obtaining causal results. Ideally, one would like to have an exogenous source of random variation in skills. However, this possibility is extremely limited in the context of international samples as here. Consequently, the present paper is mainly intended to enrich the study of the relationship between PIAAC test scores and earnings started by Hanushek et al. (2015) by providing measures of human capital that address the distinction between cognitive and non-cognitive skills.

The structure of the paper is the following. In Sect. 2 we describe the measures of non-cognitive skills that we use in our paper. Section 3 explains the empirical strategy. Section 4 describes the data. Section 5 presents the results on the association between non-cognitive skills and earnings. Section 6 shows the results after computing corrected measures for non-cognitive skills. Section 7 provides a heterogeneous analysis of the results by education, age and gender. Section 8 shows a number of robustness checks. Finally, Sect. 9 concludes.

³ See Almlund et al. (2011) for an extensive review on this evidence.

2 Non-cognitive skills

A crucial aspect of our study relies on constructing reliable measures of personality and non-cognitive skills. We use evidence from previous research to do that. Previous literature has considered self-reported and non self-reported measures based on nonresponse in a questionnaire as a proxy for non-cognitive skills. We also take advantage from the computer based nature of the PIAAC test to construct non self-reported measures based on testing behavior.

2.1 Self-reported measures of non-cognitive skills

Self-reported variables have been used extensively as the main source of information on non-cognitive skills. PIAAC study is mainly designed to obtain standardized measures of cognitive skills of adult population and, unfortunately, it provides poor information on non-cognitive skills.⁴ However, in the background questionnaire we found three items that can be related to individuals' personality and beliefs: cultural engagement, political efficacy and social trust. Participants have to give each statement a score ranking from 1 to 5. We use the following items:

Cultural engagement In the last 12 months, how often, if at all, did you do voluntary work, including unpaid work for a charity, political party, trade union or other non-profit organization? (The choice of answers is the following: "Never", "Less than once a month", "Less than once a week but at least once a month", "At least once a week but not every day", "Every day").

Cultural engagement is related to the frequency of doing voluntary work for nonprofit organizations. This indicator of volunteering from PIAAC shows lower results for subpopulations with lower literacy skills (Grotlüschen 2017). Therefore, we would expect a positive relationship between this variable and individual earnings.

Social trust There are only a few people you can trust completely (the choice of answers is the following: "Strongly agree", "Agree", "Neither agree nor disagree", "Disagree", "Strongly disagree").

Social trust has been often identified as a personality trait (see Allport 1961 or Uslaner 1999, 2000, among others). According to Uslaner (1999), social trust is related to other traits such as optimism and the control capacity. So, we expect to observe a positive relationship between this variable and the level of earnings.

Political efficacy People like me don't have any say about what the government does (the choice of answers is the following: "Strongly agree", "Agree", "Neither agree nor disagree", "Disagree", "Strongly disagree").

⁴ In the initial design, PIAAC was expected to include two specific questions addressing individuals' personality, in particular grit, self-discipline and locus of control. Unfortunately, these questions were dropped from the main study (OECD 2013).

Almond and Verba (1965), Campbell (1980) or Campbell et al. (1954) suggested that personality traits like self-esteem or personal effectiveness lies behind political efficacy. Therefore, we would expect a positive relationship between political efficacy and earnings.

Self-reports have often been subject to criticism, due to misreports, subjectivity or the socially desirable response bias studied by Paulhus (1984). Without a common metric, individual responses are not likely to be interpersonally comparable (see King et al. 2004; Soest et al. 2011). In the case of PIAAC, there is the additional problem that the above questions might provide incomplete information on individuals' non-cognitive skills.

In Sect. 8.3 and the "Online Appendix", we provide some additional measures of non-cognitive skills based on the proposal of Cabrales et al. (2014).

2.2 Non self-reported measures of non-cognitive skills

To overcome the lack of self-reported measures of non-cognitive skills, we use the PIAAC database and previous literature to construct non self-reported measures of non-cognitive skills.

Measures based on the background questionnaire

Recent research by Hedengren and Stratmann (2012), Hitt et al. (2016), Cheng et al. (2016) and Zamarro et al. (2016a, b, 2017) proposes the use of survey effort, measured by non-response rates in a questionnaire, to capture non-cognitive skills.

This approach suggests the use of non-response as a valuable proxy of socioemotional skills because of several reasons. First, the time-consuming nature of filling out a questionnaire provides a task-based measure which, differently from self-reports, is interpersonally comparable. Second, the non-challenging nature of the tasks disregards cognitive skills as determinants of non-response. Third, this information is readily available in many international data sets that do not report individual information on non-cognitive skills. As Hedengren and Stratmann (2012) indicate: "surveys contain a valuable but neglected source of data: what respondents do not say.(...). When a respondent forgets to fill answers to some questions on the survey form, or refuses to provide an answer to the interviewer, we gain important information about the respondent". Hedengren and Stratmann (2012) validate survey non-response as a measure of personality by showing that it is correlated with other traditional measures of conscientiousness and that it has predictive power on individuals' earnings and longevity. Hitt et al. (2016) show that the correlation between survey non-response and earnings is not due to simultaneity bias by using longitudinal data from NLSY 79. Zamarro et al. (2016a, 2017) further develop and validate measures of carelessness in survey response as a proxy for non-cognitive skills.

Hedengren and Stratmann (2012) found that survey non-response is both correlated with cognitive and non-cognitive skills and, consequently, it may be interpreted as a measure of the two types of skills. Indeed, as recognized by Borghans et al. (2008), separating cognitive ability from non-cognitive skills is not only empirically, but also

conceptually complicated. However, in the line of Hitt et al. (2016), we consider survey nonresponse to be a measure of non-cognitive skills rather than cognitive ability (despite being correlated with the two). Our first argument to support this interpretation is that filling out a questionnaire is a time consuming but not a mentally challenging task, which is conceptually related to personality and non-cognitive skills. Second, in our estimations, the correlation between survey non-response and PIAAC scores is relatively small (-0.09 and -0.01 for "don't know" and "refused" answers, respectively). Third, the correlation between survey non-response and IQ scores found by Hedengren and Stratmann (2012) may not be motivated by the possibility that nonresponse is measuring cognitive skills but by the opposite effect, i.e. test scores are measuring non-cognitive skills. Indeed, non-cognitive skills have been found to be important for performance on IQ tests due to the low stakes nature of the latter. As shown by Borghans et al. (2008); Duckworth et al. (2011); Gneezy and Rustichini (2000) and Segal (2012), scores of low performers on IQ tests can be improved by offering a reward. As mentioned above, some of this literature have explicitly proposed a decomposition of test scores into a cognitive and a non-cognitive component, see Borghans and Schils (2012).

We use average non-response rates in the background questionnaire that accompanies the PIAAC test as a measure of non-cognitive skills. PIAAC distinguishes between two different categories of survey non-response: "Don't know" and "Refused to answer". We compute the proportion of these two categories of answers in the questionnaire. Based on questions on individual finances, Riphahn and Serfling (2005) show that the observable characteristics of "don't know" answerers differ substantially from those of individuals providing either valid responses or refusals. This indicates that the two types of non-response might be motivated by different reasons. In a similar line, Shoemaker et al. (2002) claim that the two types of non-response should be treated differently. In particular, they show that more sensitive questions get more refusals. In contrast, the two types of non-responses appear more often for questions that require more cognitive effort to be answered. However, this association is stronger for "don't know" answers.⁵

Test-based measures

A second approach to obtain task based measures of non-cognitive skills is based on testing behavior. Achievement tests such as PISA or PIAAC assessments are intended to provide standardized measures of individuals' cognitive skills. However, as pointed out by Borghans et al. (2011), Duckworth et al. (2011) or Segal (2012), high scores in achievement tests result not only from high individual cognitive abilities but also from high non-cognitive skills such as motivation or perseverance. The low stakes nature of achievement tests puts forward non-cognitive skills, such as motivation or persistence, as an important element for performing well. This is well illustrated by Almlund et al. (2011): "A smart child unable to sit still during an exam or uninterested in exerting much effort can produce spuriously low scores on an IQ test".

⁵ In a robustness check, we sum the two categories in one variable.

Based on this idea several authors proposed the use of non self-reported measures of non-cognitive skills. One of the more promising identifications follows from Borghans and Schils (2012), who propose to use the performance drop experienced by test takers as a measure of non-cognitive skills. However, the PIAAC test has a multistage adaptive design in which an algorithm drives the participant to different exercises depending on their score on previous test items. This makes not feasible the use of the performance decline as a measure of non-cognitive skills. Moreover, as recognized by Borghans and Schils (2012), their strategy is too imprecise at the student level.

Hernández and Hershaff (2014) propose using test taking behavior as a measure for non-cognitive skills. According to them, in a test with no penalty for wrong answers and no time constraints (as PIAAC) skipping questions is not related with strategic reasons or knowledge but with a low interest to perform well in the test. In other words, when guessing has no penalty, not doing so denotes low motivation towards the test. Hernández and Hershaff (2014) show that a higher incidence of skipping questions is predictive of poorer future educational outcomes even after controlling for students' test scores. Therefore we expect the incidence of unanswered items to be negatively associated with individual earnings.

PIAAC distinguishes between two different types of non-responded items in the test: skipped items and not attempted items. Skipped items are those that the test taker does not answer after having them on the screen for more than 5 s. Not attempted items are blank questions for which the test taker spends less than 5 s on them, which means that he does not even try to solve the question.⁶ The latter includes unreached questions in those cases where respondents decided not to finish the test (even though there is no time limit). We follow the PIAAC distinction and we construct the variables "Skipped" and "Not Attempted" as the average rate of test-items in each of these two categories for every test taker in numeracy and literacy questions.

Other test-based measures

The computer-based part of the PIAAC assessment contains information that is not generally available in paper based achievement tests. This includes time spent in responding each test item or the number of actions done before item completion (e.g., clicking and typing responses). Even though these variables have not been used before in the literature, they might be useful to gain additional information on the personality of the test taker.

One can reasonably argue that in a test with no time limitation as PIAAC, time per question might be related with non-cognitive skills. We might think in two possible hypotheses. On one hand, a more motivated test taker may be willing to spend more time per question. In that case time per question should indicate higher non-cognitive skills. On the other hand, time per question could indicate some other non-cognitive skills such as ability to stay focused or grit/thrift, in which case more time per question would indicate a lower level of non-cognitive skills.

⁶ Skipped items were graded as wrong in the computation of PIAAC scores, while not attempted were graded as missing. Test takers were not aware about this distinction (OECD 2013).

We might also be especially interested in time needed for correct answers, which may indicate a more efficient use of cognitive skills. Therefore we construct two measures, average time per test-item and average time per correct test-item in numeracy and literacy tests.

The number of actions executed before completion of a test item may also be informative about (lack of) self-confidence or hesitation/resolution ability of the testtaker. We expect a higher number of actions to denote a lower level of non-cognitive skills. As in the previous case, we will distinguish between the number of actions per test-item and the number of actions needed per correct test-item in numeracy and literacy tests.

Unfortunately, the absence of specific measures of personality in the PIAAC database prevents us from reaching more precise conclusions about which traits may be captured by these additional measures provided in the computer based assessment.

As we show below, several of our test based measures (especially skipped items and non-attempted items) exhibit high correlation with PIAAC test scores. We believe that this implies that some of the results when using these variables might be quite mechanical rather than driven by non-cognitive skills. However, we show that they have explanatory power on earnings even after controlling for PIAAC test scores.

3 Empirical strategy

Our departing point is the work by Hanushek et al. (2015). In a recent paper these authors introduce PIAAC scores as a measure of human capital in a standard Mincerian equation. In particular they estimate:

$$\ln(y_i) = \alpha + \beta_1 C_i + \theta_1 S_i + \mu_1 E_i + \mu_2 E_i^2 + \mu_3 G_i + \varepsilon_i$$
(1)

where y_i is the hourly wage of individual i. S_i denotes years of schooling, E_i denotes years of labor market experience, G_i is a female indicator, and ε_i is a stochastic error term. C_i is individual's i score in the PIAAC test. The log-linear model has been a recurrent specification for the earnings equation. Heckman and Polachek (1974), specifically addressed the question of the functional form of the earnings equation finding support for the log-linear specification. Welland (1978) confirms their results when including cognitive ability in the right hand side of the equation.

Hanushek et al. (2015) show that PIAAC scores are positively associated with individual earnings. They interpret this finding as evidence on the importance of cognitive skills for individual earnings. However, there is a growing literature that highlights that scores in low stakes tests such as PIAAC are largely influenced by non-cognitive skills such as motivation. A natural question arising from these results is how the association between internationally standardized test scores and individual earnings (Hanushek et al. 2015) can be attributed to the cognitive or the non-cognitive components of human capital.

To answer this question we extend the model estimated by Hanushek et al. (2015) to include different measures of non-cognitive skills. Therefore, we estimate:

$$\ln(y_i) = \alpha + \beta_1 C_i + \beta_2 NC_i + \theta_1 S_i + \mu_1 E_i + \mu_2 E_i^2 + \mu_3 G_i + \varepsilon_i$$
(2)

where NC_i denotes non-cognitive skills, measured by the variables described above. NC_i may be a variable or a vector of variables depending on whether we include one or many of the measures for non-cognitive skills in the regression.

Secondly, we propose an alternative path to study how non-cognitive skills may affect the previous estimates of the relationship between earnings and cognitive skills found by Hanushek et al. (2015). According to Borghans et al. (2011), Duckworth et al. (2011) and Segal (2012) non-cognitive skills can affect PIAAC scores directly through testing behavior. However, there is a second channel through which non-cognitive skills can affect PIAAC test scores which is the formation of cognitive skills. According to Cunha and Heckman (2008) and Cunha et al. (2010), non-cognitive skills play an important role in the development of cognitive skills while the same is not true in the other way around. This second channel is more subtle and more difficult to detect as it is directly embedded in the cognitive skills accumulated by test takers. However, we might be interested in trying to have an idea on the importance of this "formation" role of non-cognitive skills. To do so we implement the following two-stage estimation of Eq. (1). In an initial step we regress:

$$C_i = \lambda N C_i + v_i \tag{3}$$

And then we use the residuals of this estimation \hat{v}_i as a corrected measure of cognitive skills in Eq. (1), that is:

$$\ln(\mathbf{y}_{i}) = \alpha + \beta'_{1}\hat{v}_{i} + \theta_{1} \mathbf{S}_{i} + \mu_{1} \mathbf{E}_{i} + \mu_{2} \mathbf{E}^{2}_{i} + \mu_{3} \mathbf{G}_{i} + \varepsilon_{i}$$
(4)

The objective of this correction is to use in the Mincerian equation estimated by Hanushek et al. (2015) only the variation in cognitive skills that is orthogonal to non-cognitive skills. Grönqvist et al. (2017) implement a similar correction, but reversing the dependent and explanatory variables.⁷ Interpretation of these results should be made with caution especially when using test based measures of non-cognitive skills. There are obvious concerns arising from using this type of correction. First, note that if the proposed measures of non-cognitive skills are also capturing cognitive skills, as it is very likely for test based measures, we will be overcorrecting the measures of cognitive skills.

As we show in our results, cognitive skills maintain its prominent role in the relationship with earnings. Thus, despite the potential overcorrection that we might be implementing, our results reinforce the importance of cognitive skills in the earnings equation. This (partially) mitigates the concern raised by the literature on non-cognitive skills towards the possibility that the role of cognitive skills have been overestimated.

⁷ In their study, they analyze the intergenerational transmission of non-cognitive skills. In the absence of measures of parental non-cognitive skills, they use schooling information (more precisely, grades on non-theoretical subjects as well as survey information on educational aspirations and social interactions) to construct such a measure. As this measure is very likely to be also capturing parents' cognitive ability, they regress it on the measure of parental test scores to net out any cognitive ability captured by this schooling information.

4 Data

In its first round, PIAAC provides internationally comparable data about skills of adult population (16–65 years) in 24 countries. These countries are Australia, Austria, Belgium (albeit just Flanders), Canada, Cyprus, the Czech Republic, Denmark, Estonia, Finland, France, Germany, Ireland, Italy, Japan, Korea, the Netherlands, Norway, Poland, the Slovak Republic, Spain, Sweden, the Russian Federation, the United Kingdom (specifically England and Northern Ireland), and the United States. In our study, we exclude countries for which continuous data on earnings is not available: Austria, Canada, Germany, Sweden and the United States. Russian Federation (OECD 2013). Australia and Cyprus are not included because data is not publicly available. Therefore, we are left with 16 countries in our estimations.

PIAAC is intended to measure cognitive skills in three different domains: numeracy, literacy and problem solving. In the present paper we follow Hanushek et al. (2015) and consider the numeracy domain as a measure for cognitive skills.⁸ Problem solving domain was optional and Cyprus, France, Italy and Spain did not take that option. Moreover, problem solving involves longer questions and less test items, which may affect some of our measures such as time per question or the number of skipped items.

Skills are measured preferably with a computer-based multiple item adaptive test, which drives test takers through different test paths depending on their performance on previous test items. The advantage of the computer-based assessment is that it offers rich information on test performance, in particular time used in each test item or the number of actions before completing a test item (e.g., clicking and typing responses). Therefore, in order to use this additional information and to have comparable subsamples, we restrict our analyses to the computer-based assessment. PIAAC assessment also gathers information about respondents labor market status, education, earnings, experience, socio-demographic characteristics and attitudes towards learning by means of a background questionnaire filled in before doing the test. The assignment of test takers to the computer-based test is clearly non-random as it depends on having previous experience with computers as well as on successful performance on an ICT test (previous to the assessment). Thus, our results cannot be extended to the group that did the paper based assessment. According to the PIAAC technical report (OECD 2016), on average a 73.5% of the respondents took the computer based form of the assessment.

Moreover, the adaptive multistage design of the computer based assessment might have some implications in the computation of test based measures of non-cognitive skills. In particular, given that best performers are driven to more difficult questions, we might expect that time per question, time per correct answer and the number of non-attempted or skipped items to be higher for more difficult items.⁹

⁸ Our results are robust to consider literacy or problem solving domains instead of numeracy. These results are available in the "Online Appendix" and briefly commented in Sect. 8.6.

⁹ The consequences that the adaptive design may have on actions per test item and on actions per correct answer are unclear.

We follow Hanushek et al. (2015) to obtain the variables included in Eq. (1). Therefore, we use the logarithm of gross hourly wages as the dependent variable of the model.

In PIAAC, skills are a latent variable that is estimated using item-response-theory models (see OECD 2013 for details). PIAAC provides 10 plausible values, instead of only one individual score, for each respondent and each skill domain. Using the average of the 10 plausible values provides an unbiased estimate of individual skills in each domain.

Years of schooling is measured as the number of years of education equivalent to the highest level of education obtained. Experience is measured as actual work experience, indicating the number of years the person has been working at least 6 months in full-time or part-time work.¹⁰ As mentioned in Hanushek et al. (2015), whose analysis is by country, there is considerable variation across countries in average years of schooling and actual work experience in the sample of full-time prime-age workers.

Given the stratified and clustered nature of the PIAAC sample, we use PIAAC weights in all our regressions to ensure that national samples are representative of each country population. Furthermore, because of the complex sample design in PIAAC, the standard errors are computed using the replication method. For PIAAC research, the so called *jacknife* replicate procedure was chosen (for more detail see Efron 1982 or Levy and Lemeshow 1999).¹¹

We use country fixed effects in all our pooled regressions to account for country idiosyncratic characteristics.

With the aim of obtaining a homogenous sample with strong labor-force commitment we also follow Hanushek et al. (2015) in limiting the sample to prime age non self-employed full-time workers. These are defined as workers between 35 and 54 years of age and working at least 30 h/week. As Hanushek et al. (2015) point out, by doing this we avoid the influence of some factors such as family demands or health limitations. However, we might be obtaining lower estimates for skills as long as higher skills also lead to higher employability. Finally, to avoid the influence of outliers we also trim the top and bottom 1% of the wage distribution of each country to prevent the influence of outliers.^{12, 13}

Table 1 provides descriptive statistics of the variables that we use in our estimations. PIAAC scores are measured on a 500-point scale. To simplify the interpretation, PIAAC scores as well as all our measures for non-cognitive skills are standardized to

¹⁰ Actual work experience could be endogenous to the skills variable and one could argue that experience is actually one of the channels by which skills could affect earnings. Therefore, as in Hanushek et al. (2015), we perform our main estimations by using potential work experience calculated as age minus years of schooling minus six. In Sect. 8.2 we discuss again this issue. Results do not change significantly and they are available upon request.

¹¹ The correct estimation of standard errors with PIAAC data is done with the Stata command, *repest* (see Stata help for more details).

¹² Hanushek et al. (2015) suggest that the first source for outliers seems to be errors in data entry or coding.

¹³ We also restricted the sample to only native-language speakers (about 95% of our sample), as noncognitive test based measure such as average time per test item or proportion of skipped items could be affected by the language of assessment. Results hardly change and are available upon request.

Table 1 Descriptive statistics

Variable	Mean	SE	Min.	Max.
Gross hourly wage in PPP US\$	18.06	9.65	1.91	105.35
PIAAC score in numeracy	287.01	44.86	104.83	454.13
Years schooling	13.50	2.83	3.00	22.00
Experience (years)	20.97	7.55	0.00	50.00
Female (share)	0.39	0.49	0.00	1.00
Non-cognitive skills test based measu	res			
Time per test item (s)	73,572.70	81,047.57	5456.76	6,372,903.00
Time per correct test item (s)	67,048.45	28,130.70	6743.38	619,218.44
Skipped	0.05	0.10	0.00	1.00
Not attempted	0.02	0.07	0.00	1.00
No. of actions per test item	2.69	1.35	0.00	55.10
No. of actions per correct test item	2.57	1.46	1.00	51.00
Non-cognitive skills measures from be	ackground quest	tionnaire		
Don't know	0.0018	0.0051	0.0000	0.0800
Refused to answer	0.0004	0.0021	0.0000	0.0408
Self-reported non-cognitive skills				
Cultural engagement	1.58	0.98	1.00	5.00
Political efficacy	2.68	1.24	1.00	5.00
Social trust	2.30	1.11	1.00	5.00
Observations	14,591			

Observations per country: Belgium (899), Czech Republic (657), Denmark (1.392), Estonia (1032), Finland (1095), France (1280), Ireland (662), Italy (708), Japan (831), Korea (867), Netherlands (800), Norway (1152), Poland (334), Russian Federation (352), Slovak Republic (645), Spain (893), United Kingdom (1344)

have a zero mean and a standard deviation equal to one with respect to the pool of countries that we use in our estimations.

5 Results: non-cognitive skills in a Mincerian equation

We focus our estimations on the pooled sample containing all individuals of all countries with available information on earnings.

5.1 Non self-reported measures of non-cognitive skills using the PIAAC database

Our non self-reported measures of non-cognitive skills are test-based measures as well as non-response in the background questionnaire.

In Table 2, we show the correlations between PIAAC scores and the different proxies for non-cognitive skills. As expected, skipped and not-attempted questions present a

Table 2 Correlation matrix between PIAAC score in numeracy and non-cognitive skills measures	ween PIAAC	score in nun	neracy and n	on-cognitiv	e skills meas	ures						
	PIAAC score in numeracy	Time per test item	Time per correct test item	Skipped	Not attempted	No. of actions per test item	No. of Don't actions per know correct test item	Don't know	Refused to Cultural answer engage- ment	Cultural engage- ment	Political efficacy	Social trust
PIAAC score in Numeracy	-											
Time per test item	0.06	1										
Time per correct test item	0.21	0.32	1									
Skipped	-0.52	-0.04	-0.11	1								
Not attempted	-0.25	0.11	-0.08	0.40	1							
No. of actions per test item	0.12	0.11	0.32	-0.27	-0.22	1						
No. of actions per correct test item	m 0.05	0.09	0.30	-0.07	-0.03	0.78	1					
Don't know	-0.09	0.01	0.03	0.07	0.04	-0.02	0.00	1				
Refused to answer	-0.01	-0.01	-0.01	0.02	0.05	-0.02	-0.01	0.01	1			
Cultural engagement	0.10	-0.01	0.00	-0.07	-0.03	-0.01	-0.02	- 0.02 -	-0.03	1		
Political efficacy	0.18	0.01	0.02	-0.14	-0.06	0.04	0.02	-0.05	0.01	0.07	1	
Social trust	0.15	-0.01	0.01	-0.12	-0.04	0.02	0.00	- 0.04 -	-0.01	0.10	0.30	1
Time per test item: average time spent in numeracy and literacy tests, time per correct test item: average time spent in responding correct answers in numeracy and literacy tests, skipped: proportion of skipped/no response questions in the in the numeracy and literacy tests (more than 5"), not attempted: proportion of not reached/not attempted questions in the in the numeracy and literacy tests (more than 5"), not attempted: proportion of not reached/not attempted questions in the in the numeracy and literacy tests, No. of actions per test item: average number of actions in the numeracy tests, No. of actions per test item: average number of actions in the numeracy and literacy tests, No. of actions per correct test item: average number of actions in the numeracy tests, No. of actions per correct test item: average number of actions in the numeracy tests, No. of actions per correct test item: average number of actions in the numeracy tests, No. of actions per correct test item: average number of actions in the numeracy tests, No. of actions per correct test item: average number of actions in the numeracy tests, No. of actions per test item: average number of actions in the background questionnaire. For test item: average number of actions in the background questionnaire, refused answer: proportion of "refused" answers in the background questionnaire.	spent in nume to response qui cy tests, No. c cy and literacy ound questionr	racy and lite estions in the of actions per <i>v</i> tests, don't naire	racy tests, tir e in the nume r test item: ar know answe	me per corre eracy and lit verage numb er: proportio	ent in numeracy and literacy tests, time per correct test item: average time spent in responding correct answers in numeracy and literacy tests, response questions in the in the numeracy and literacy tests (more than 5"), not attempted: proportion of not reached/not attempted questions tests. No. of actions per test item: average number of actions in the numeracy and literacy tests, No. of actions per correct test item: average and literacy tests, don't know answer: proportion of "Don't Know" answers in the background questionnaire, refused answer: proportion of d questionnaire	verage time nore than 5", in the nume Know" answ	spent in resp.), not attemp racy and lite ers in the ba	oonding c ted: prop racy tests ckground	orrect answortion of no , No. of act questionna	ers in numer t reached/no ions per corr ire, refused a	acy and liter t attempted ect test item answer: prop	acy tests, questions : average ortion of

negative notable correlation with PIAAC proficiency, as they are directly based on test items. Time per correct test item also shows an important positive correlation with PIAAC scores. As we can see the remaining measures show a low correlation with PIAAC scores.

In Table 3 we present the results of estimating Eqs. (1) and (2). In the first three columns we can see the estimates for Eq. (1): first with only years of schooling as a measure of human capital, second with only PIAAC scores, i.e., the score in numeracy and third with years of schooling and PIAAC scores together. Even though our pooled sample includes fewer countries than the one in Hanushek et al. (2015), our estimates do not present significant differences from theirs. One standard deviation increase in PIAAC numeracy scores is associated with 18.7% higher earnings and 10% after controlling for years of schooling.¹⁴ By comparing columns (1) and (3) we observe that the coefficient of years of schooling decreases by a 16.4% after including numeracy test scores. Hanushek and Woessmann (2008) attribute the reduction experienced in the coefficient of years of schooling together with the larger value of the adjusted R^2 in column (2), to the better measure of human capital provided by test scores in comparison to years of schooling. An alternative interpretation of this finding is in terms of returns to schooling and ability bias. If we interpret θ_1 as returns to schooling, the observed reduction might be associated to the ability bias, see Card (1999) among others.¹⁵

In the subsequent columns we include our constructs for non-cognitive skills. In columns (4), (6) and (8) proxies for non-cognitive skills are included in separate regressions while in columns (5), (7) and (9) we include all of them as explanatory variables in the same regression.

We estimate first the effects of non-cognitive skills without controlling for the measures of human capital, i.e., years of schooling and PIAAC scores. We add controls for PIAAC score in numeracy test in columns (6) and (7) as well as years of schooling in columns (8) and (9).

Our estimates are consistent with previous studies that proposed survey nonresponse as a measure of non-cognitive skills. As in Hedengren and Stratmann (2012) and Hitt et al. (2016) the ratio of "don't know" answers in the questionnaire is negatively associated with individual earnings. When not controlling for PIAAC scores and level of education, we find that one standard deviation in questionnaire non-response, is associated with 8.54% lower earnings. We complement the findings of Hedengren and Stratmann (2012) and Hitt et al. (2016) by showing that this result is robust to controlling for PIAAC test scores. When controlling for PIAAC scores and level of education, one standard deviation in questionnaire non-response, is associated with a 3.58% reduction in earnings. We also see that the different types of non-response, i.e., "don't know" and "refused to answer", have opposite effects (also when included in separate regressions), even though the proportion of "refused to answer" is not

¹⁴ The (small) differences arising with respect to the estimates in Hanushek et al. (2015) are completely due to the fact that we only have available data for a smaller set of participating countries.

¹⁵ In contrast, the coefficient of years of schooling does not change when we control for measures of non-cognitive skills but not for cognitive skills. By doing this exercise, the value of θ_1 is very similar to the one estimated in column (1) of Table 3 (its value varies from 0.724 to 0.768).

Table 3 Non-cognitive skills in the Mincerian equation (non-self reports)	in the Mincer	rian equation	n (non-self re	ports)					
	(1)	(2)	(3)	(4) Separated	(5) Joint	(6) Separated	(7) Joint	(8) Separated	(9) Joint
PIAAC score in numeracy		0.187***	0.187*** 0.103*** (0.007) (0.008)			0.177***_0.195***	0.184***	0.0982***_0.110***	0.106***
Years of schooling	0.0767***		0.0641***					(0.002)***_0.0642*** (0.002)	
Non-cognitive skills test based measures	measures								
Time per test item				-0.00285	-0.00277 **	-0.00782	-0.00264	-0.00472	-0.000433
				(0.002)	(0.001)	(0.005)	(0.002)	(0.004)	(0.002)
Time per correct test item				-0.00797	-0.0115*	-0.0403^{***}	-0.0379^{***}	-0.0331^{***}	-0.0321^{***}
				(0.007)	(0.007)	(0.007)	(0.008)	(0.006)	(0.007)
Skipped				-0.0897^{***}	-0.109^{***}	-0.0175^{**}	-0.0309**	-0.00904	-0.0158^{**}
				(0.006)	(0.008)	(0.007)	(0.008)	(0.007)	(0.008)
Not attempted				-0.0211^{***}	0.0168^{**}	0.0222^{***}	0.0289^{***}	0.0149*	0.0189*
				(0.007)	(0.008)	(0.007)	(600.0)	(0.008)	(0.011)
No. of actions/test item				0.00612	-0.0306^{***}	-0.0101^{**}	-0.00968	-0.00841*	-0.00403
				(0.006)	(0.00)	(0.005)	(0.008)	(0.005)	(0.008)
No. of actions/correct test item	-			-0.000352	0.0211***	-0.00530	0.0103	-0.00585	0.00423
				(0.005)	(0.008)	(0.004)	(0.007)	(0.004)	(0.007)
Non-cognitive skills measures from Background questionnaire	from Backgro	ound questio	nnaire						
Don't know answer				-0.0854^{***}	-0.0770^{***}	-0.0569^{***}	-0.0541^{***}	-0.0358^{**}	-0.0337^{**}
				(0.015)	(0.015)	(0.014)	(0.014)	(0.014)	(0.014)

Table 3 continued

	(1)	(2)	(3)	(4) Separated	(5) Joint	(6) Separated	(7) Joint	(8) Separated	(9) Joint
Refused to answer				0.0544	0.0570*	0.0613*	0.0514	0.0190	0.0126
				(0.034)	(0.034)	(0.033)	(0.033)	(0.029)	(0.029)
Adj. R2		0.361	0.456	0.272 - 0.301	0.306	0.359-0.361	0.368	0.455-458	0.459
No. of observations		14,591	14,523	14,450–14,523	14,518	14,450–14,523	14,518	14,450–14,523	14,450
Dependent variable: log gross hour	gross hou	Irly wages. Esti	mations using s	y wages. Estimations using sample weights and 80 replicates for standard errors. All estimations include a constant, experience, the square	replicates for st	andard errors. All estin	nations include	a constant, experience	, the square

of experience, a dummy for female and country fixed effects Standard errors in parentheses * p < 0.10, ** p < 0.05, *** p < 0.01 significantly correlated with earnings. The existence of differences in the "refused" and "don't know" categories of non-response has been emphasized by Riphahn and Serfling (2005) and Shoemaker et al. (2002). If we collapse the two measures in a single variable of non-response, the results are similar to the ones obtained for "don't know" answers.¹⁶ Thus, consistently with Hedengren and Stratmann (2012) and Hitt et al. (2016) when using a unique category, survey non-response is negatively correlated with earnings. Moreover, our results suggests that this relationship is exclusively driven by a specific type of non-response: "don't know" answers.

Some of the test based measures are significantly associated with individual earnings in most of the specifications. Time per correct answer and skipped test items are significantly and negatively associated with individual earnings, respectively. The size of these coefficients implies that one standard deviation decrease in the average time spent per correct answer or in the incidence of skipping test items is associated with 3.21 and 1.58% higher earnings, respectively, after controlling for PIAAC score in numeracy and number of years of schooling [column (9) of Table 3]. The size of these estimates is relatively lower than that of PIAAC scores.; however we need to have in mind that these may be interpreted as lower bounds since PIAAC scores are likely to be also influenced by non-cognitive skills.¹⁷

Skipped test items are negatively associated with individual earnings in all specifications, confirming the findings by Hernández and Hershaff (2014) [except when controlling for PIAAC scores and years of education in column (8), when the result is not statistically significant]. Conditional on having the same level of cognitive test scores, individuals that do not answer questions in a non-penalized test have lower earnings.

Average time per correct test item has a negative and statistically significant coefficient in the specifications of Table 3 [except in column (4) where we do not control for cognitive skills and it is included separately in the regression]. Having in mind that this variable is also relevant after controlling for PIAAC scores, this means that, conditional on having the same proficiency in PIAAC numeracy test, needing less time to provide the correct answer for a question is positively associated with earnings. This finding suggests the idea that human capital does not only involve having numeracy skills but also being able to use them in an efficient way.

Unstable results arise for not attempted test items. Not attempted items are significantly negatively associated with individual earnings when we do not control for cognitive test scores, while they are positively associated with individual earnings when we control for cognitive test scores. We suspect that this result is mainly driven by reversed causality. People with a higher opportunity cost (higher earnings) might be less willing to spend time in the test. An alternative explanation for the change in the coefficient of non-attempted items might arise from the multi-adaptive design of the computer based assessment discussed above. Better performers might be driven to more difficult questions, which in turn might be more likely to be left non-attempted.

¹⁶ Results are available upon request.

¹⁷ This might be especially relevant in the case of rate of skipped questions, which exhibits substantial correlation with PIAAC scores but it does not decrease the size of PIAAC coefficients when jointly added in the regression.

Despite low performers exhibit a higher overall level of non-attempted items (the correlation is -0.25, see Table 2), the effect of the adaptive design on increasing the number of non-attempted questions may be captured only when controlling for test performance. This explains why the coefficient of non-attempted items is only negative when not including further controls and positive in all other cases.

When using the additional test based measures we find that average time per question is not significantly correlated with individual earnings (except when included jointly with other non-self-reported measures and not controlling for PIAAC test scores [column (5)].

"Number of actions" has a negative effect in some of the specifications, while the coefficients of "number of actions per correct test items" are more unstable across our specifications. The potential collinearity between "number of actions per correct answer" and "number of actions" (from Table 2, the correlation coefficient between these two variables is the highest: 0.78) may cause that the coefficient of the former changes its sign when both of them are included in a joint regression.

5.2 Self-reported measures of non-cognitive skills using the PIAAC database

Table 4 reports the results of estimating Eq. (2) using as measures for non-cognitive skills the self-reported variables. As in Table 3, columns (1), (3) and (5) show the estimations by introducing separately these three measures, while in columns (2), (4) and (6) we include all of them in the estimations jointly with the non self-reported measures from the previous sub-section. In columns (1) and (2) we do not control for the traditional measures of cognitive skills, in columns (3) and (4) we control for PIAAC scores in numeracy and in columns (5) and (6) we add years of schooling.

Political efficacy and social trust have statistically significant coefficients in all the specifications, when they are included both separately and jointly. Both of them are positively related to earnings, as expected. A one standard deviation increase in political efficacy or in social trust is associated with 1.62 and 2.31% higher earnings, when we control for both the PIAAC score in numeracy and the number of years of schooling. The magnitude of these effects is similar to the impact of time per correct test item or of skipped test items.

The frequency of cultural engagement is positively related to earnings in all columns, but it is statistically significant only in columns (1) and (2), when we do not control neither for PIAAC scores nor for years of schooling.

When we include both self-reported and non-self-reported measures of noncognitive skills in the estimations, we find robust significant results for the same variables as in Table 3: "don't know" answers, skipped test items and non-attempted items, as well as time per correct answer in columns (4) and (6).

6 Results: corrected measures of cognitive skills

As we have previously commented, there is a growing attention towards the importance of non-cognitive skills in the production of tests' scores. On the one hand, the low stakes nature of the PIAAC test may imply that personality plays a prominent role

Table 4 Non-cognitive skills in the Mincerian equation (self reports)	in the Mincerian equ	tation (self reports)				
	(1) Separated	(2) Joint	(3) Separated	(4) Joint	(5) Separated	(6) Joint
PIAAC score in numeracy			0.180***_0.186***	0.174***	0.101***_0.103***	0.103***
Years of schooling			(000.0)	(600.0)	(0.0625***_0.0641***	0.0611***
					(0.002)	(0.002)
Self-reported non-cognitive skill measures	ill measures					
Cultural engagement	0.0244^{***}	0.00984^{*}	0.00893	0.00298	0.000511	-0.00241
	(0.005)	(0.005)	(0.006)	(0.005)	(0.005)	(0.005)
Political efficacy	0.0612^{***}	0.0406^{***}	0.0410^{***}	0.0292^{***}	0.0217^{***}	0.0162^{***}
	(0.006)	(0.006)	(0.006)	(0.006)	(0.005)	(0.006)
Social trust	0.0676***	0.0485^{***}	0.0505***	0.0414^{***}	0.0271^{***}	0.0231^{***}
	(0.006)	(0.007)	(0.006)	(0.007)	(0.006)	(0.006)
Non self-reported non-cognitive skill measures	ve skill measures					
Time per test item		-0.00216*		-0.00215		-0.000313
		(0.001)		(0.002)		(0.002)
Time per correct test item		-0.0102		-0.0354^{***}		-0.0308^{***}
		(0.007)		(0.008)		(0.006)
Skipped		-0.0997^{***}		-0.0279^{***}		-0.0150*
		(0.008)		(0.008)		(0.008)

ര്
E
·=
Ħ
continu
~ ~
_
-
4
4
4
4
4
4

	(1) Separated	(2) Joint	(3) Separated	(4) Joint	(5) Separated	(6) Joint
Not attempted		0.0161^{*}		0.0277 ***		0.0189*
		(0.008)		(6000)		(0.011)
No. of actions/test item		-0.0283^{***}		-0.00922		-0.00399
		(0.008)		(0.008)		(0.008)
No. of actions/correct test item		0.0194^{***}		0.00965		0.00385
		(0.007)		(0.007)		(0.006)
Don't know answer		-0.0785^{***}		-0.0569^{***}		-0.0363**
		(0.015)		(0.015)		(0.015)
Refused to answer		0.0584		0.0527		0.0185
		(0.036)		(0.035)		(0.031)
Adj. R2	0.272 - 0.287	0.325	0.361 - 0.360	0.380	0.456-459	0.463
Observations	14,549-14,588	14,471	14,549–14,588	14,471	14,482-14,520	14,404
Dependent variable: log gross hourly wages. Estimations using sample weights and 80 replicates for standard errors. All estimations include a constant, experience, the square	wages. Estimations using	g sample weights and 8	0 replicates for standard	errors. All estimations	include a constant, experi	ience, the square

of experience, a dummy for female and country fixed effects Standard errors in parentheses * p < 0.10, ** p < 0.05, *** p < 0.01

in obtaining high scores in the test. On the other hand, non-cognitive skills play an important role in the formation of cognitive skills while the same is not true in the other way around (Cunha and Heckman 2008; Cunha et al. 2010). If that is the case, the estimates of Hanushek et al. (2015) may be better interpreted as the effect of a broader notion of human capital rather than the specific effect of cognitive skills on earnings. In other words, the relationship between PIAAC scores and earnings might be mediated by non-cognitive skills. We might be interested in studying how this may affect the highly studied association between earnings and cognitive skills.

Table 5 presents the results obtained after implementing the correction proposed in Eqs. (3) and (4). Notice that, by doing so we are only using the part of cognitive skills that is orthogonal to our proxies for non-cognitive skills.

Similarly, in Table 6 we use self-reported measures to apply the same correction.

We can see that in all cases, the estimated coefficients for the corrected measures of PIAAC scores are smaller than the uncorrected ones. In most of the cases the variation is higher than 12%. The highest reduction takes places when using skipped questions to correct for the effect of non-cognitive skills (-35.63%, controlling for years of education). This is not surprising as skipping incidence is the proxy for non-cognitive skills that presents the highest correlation with PIAAC scores.

Due to the high correlation of some of the items we might be overcorrecting PIAAC test scores, which might downward bias the estimate of β_1 . However, test scores are still highly significant after the correction. This finding supports the prominent role of cognitive skills in the earning equation and reinforces the results of Hanushek et al. (2015). This (partially) mitigates the concern raised by the literature on non-cognitive skills towards the possibility that the role of cognitive skills has been overestimated.

7 Subgroup analysis

7.1 Differences by education level

In this sub-section we estimate the impact of non-cognitive skills separately for different education levels. We distinguish three levels of education: low education (lower secondary, primary or less education), medium education (upper-secondary, postsecondary and other non-tertiary education) and higher education (tertiary education). Results of estimations of specifications (1) and (2) are shown in Table 7.

We observe that, in general, there are no significant changes in the sign and the statistical significance of our variables of interest in all the three groups. However, there are some differences among the different measures of non-cognitive skills and according to the different educational levels.

Skipped test items and "don't know" answers have a negative and statistically significant effect on earnings regardless of the level of education.

In the case of skipped questions and when we do not control for cognitive skills, our results show that the higher the educational attainment of the individual, the stronger the impact of non-cognitive skills. The estimates almost double, when we compare a low-educated person with a high-educated person: 5.63 versus 9%. However, when we control for cognitive skills measured by the PIAAC score and years of schooling, we

Table 2 Collected Incastics for cognitive sums (non seni-tepores)		INC HOLL CHING	(e)10da1-1						
	Time per test item	Time per correct test item	Skipped	Not attempted	No. of actions/test item	No. of actions/correct test item	Don't know answer	Refused to answer	All Non-cognitive skills test items
Corrected PIAAC score	0.161^{***} (0.006)	0.164^{***} (0.006)	0.128*** (0.007)	0.162^{***} (0.006)	0.161^{***} (0.006)	0.161^{***} (0.006)	0.185^{***} (0.007)	0.187^{***} (0.007)	0.132*** (0.006)
Years of schooling									
Adj. R2	0.361	0.363	0.328	0.361	0.36	0.36	0.359	0.361	0.332
No. of obs.	14,568	14,518	14,591	14,591	14,590	14,518	14,591	14,591	14,518
Corrected PIAAC score	0.0888^{***}	0.0921^{***}	0.0663^{***}	0.0897^{***}	0.0890^{***}	0.0884^{***}	0.102^{***}	0.103^{***}	0.0709^{***}
	(0.007)	(0.007)	(0.007)	(0.007)	(0.007)	(0.007)	(0.008)	(0.008)	(0.006)
Years of schooling	0.0640^{***}	0.0636^{***}	0.0691^{***}	0.0641^{***}	0.0641^{***}	0.0641^{***}	0.0643***	0.0641^{***}	0.0685^{***}
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)
Adj. R2	0.456	0.457	0.447	0.456	0.456	0.455	0.455	0.456	0.448
Observations	14,500	14,450	14,523	14,523	14,522	14,450	14,523	14,523	14,450
Estimations of log gross hourly wage on corrected PIAAC score in Numeracy. Each column shows only the coefficient of corrected PIAAC score when correcting for each of the non-cognitive skills measures that we use in the paper. Estimations using sample weights and 80 replicates for standard errors. All estimations include a constant, experience, the square of experience, a dummy for female and country fixed effects Standard errors in parentheses * $p < 0.10$, *** $p < 0.05$, *** $p < 0.01$	ourly wage on () measures that experience, a du eses $* p < 0.10$	corrected PIAA we use in the F mmy for female $, ** p < 0.05$,	C score in Num apper. Estimatic a and country fi **** $p < 0.01$	teracy. Each coll ins using sample xed effects	umn shows only e weights and 8	y the coefficient of o 0 replicates for star	corrected PIAA ndard errors. A	C score when co Il estimations in	orrecting for each clude a constant,

 Table 5
 Corrected measures for cognitive skills (non self-reports)

SERIEs (2017) 8:417-473

	Cultural Political engagement efficacy	Political efficacy	Social trust	All self-reported non-cognitive skills	All self-reported and non- self-reported non-cognitive skills	Cultural engagement	Political efficacy	Social trust	All self-reported non-cognitive skills	All self-reported and non-self- reported non-cognitive skills
Corrected 0.181** PIAAC score (0.007)	0.181^{***} (0.007)	0.169*** (0.007)	0.171^{***} (0.008)	0.159*** (0.007)	0.121^{***} (0.006)	0.0998*** (0.008)	0.0910^{***} (0.007)	0.0919^{***} (0.008)	0.0848^{***} (0.007)	0.0641^{***} (0.006)
Years of schooling						0.0648*** (0.002)	0.0663*** (0.002)	0.0662*** (0.002)	0.0676*** (0.002)	0.0700*** (0.002)
Adj. R2 No. of obs.	0.356 14,588	0.347 14,549	0.348 14,582	0.340 14,542	0.323 14,471	0.455 14,520	0.452 14,482	0.452 14,514	0.451 14,475	0.446 14,404
Estimations of log gross hourly wage on corrected PIAAC score in Nu of the non-cognitive skills measures that we use in the paper. Estimat experience, the square of experience, a dummy for female and country Standard errors in parentheses * $p < 0.10, ** < p < 0.05$	log gross hourl nitive skills me square of experin parentheses	y wage on co assures that w rience, a dum * p < 0.10,	prected PIAAC e use in the pa my for female ** $p < 0.05$, **	Estimations of log gross hourly wage on corrected PIAAC score in Numeracy. Eacl of the non-cognitive skills measures that we use in the paper. Estimations using scenteince, the square of experience, a dummy for female and country fixed effects Standard errors in parentheses * $p < 0.10$, ** $p < 0.05$, *** $p < 0.01$	Estimations of log gross hourly wage on corrected PIAAC score in Numeracy. Each column shows only the coefficient of corrected PIAAC score when correcting for each of the non-cognitive skills measures that we use in the paper. Estimations using sample weights and 80 replicates for standard errors. All estimations include a constant, experience, the square of experience, a dummy for female and country fixed effects Standard errors in parentheses * $p < 0.10$, *** $p < 0.05$, *** $p < 0.01$	lows only the cc its and 80 repli	befficient of cor cates for standa	rected PIAAC ard errors. All (score when corrected	cting for each le a constant,

 Table 6
 Corrected measures for cognitive skills (self-reports)

 ${\textcircled{2}} {\underline{\bigcirc}} Springer$

	(1)	(2)	(3)	(4)	(5)	(9)	(7) Separated	(8)	(6)	(10) Joint	(11)	(12)
	Low educ.	Medium educ.	High educ.	Low educ.	Medium educ.	High educ.	Low educ.	Medium educ.	High educ.	Low educ.	Medium educ.	High educ.
PIAAC score in numeracy	0.0768***	0.0768*** 0.0976***	0.1508***	0.0728***	0.0915***	0.1191***						
	(0.022)	(0.011)	(0.013)	(0.023)	(0.011)	(0.013)						
Years of schooling				0.0137	0.0492***	0.0824***						
				(0.012)	(0.010)	(0.006)						
Non-cognitive test based measures	ısed measures											
Time per test item							-0.0002	0.0015	-0.0348^{***} 0.0740	0.0740	0.0010	-0.0111
							(0.011)	(0.001)	(0.011)	(0.092)	(0.001)	(0.008)
Time per correct test							-0.0053	-0.0060	-0.0239^{***}	-0.0281	-0.0106	-0.0234^{**}
item							(0.018)	(0.010)	(600.0)	(0.042)	(0.011)	(0.010)
Skipped							-0.0370^{***}	-0.0512^{***}	-0.0762^{***}	-0.0563^{***}	-0.0589^{***}	-0.0899^{***}
							(0.013)	(0.007)	(0.011)	(0.015)	(0.009)	(0.015)
Not attempted							0.0104	-0.0145*	-0.0184*	0.0340	0.0071	0.0057
							(0.019)	(600.0)	(0.010)	(0.028)	(0.009)	(0.013)
No. of actions/test							0.0064	0.0051	-0.0020	-0.0267	-0.0138	-0.0173
item												
							(0.014)	(0.007)	(0.007)	(0.019)	(0.014)	(0.014)

	(1)	(2)	(3)	(4)	(5)	(9)	(7) Separated	(8)	(6)	(10) Joint	(11)	(12)
	Low educ.	Medium educ.	High educ.	Low educ.	Medium educ.	High educ.	Low educ.	Medium educ.	High educ.	Low educ.	Medium educ.	High educ.
No. of actions/correct test	t.						0.0029 (0.015)	(0.009) – 0.0009 – 0.008)	- 0.0044 (0.004)	0.0161 (0.019)	0.0083 (0.013)	0.0128 (0.011)
uem Non-cognitive measures from Background questionnaire	ures from Ba	ckground quest	tionnaire									
Don't know answer	ı						-0.0548	-0.0453^{**}	-0.0527^{**}	-0.0639*	-0.0379*	-0.0471^{**}
							(0.041)	(0.021)	(0.022)	(0.038)	(0.022)	(0.022)
Refused to answer							-0.0648	-0.0056	0.0549	-0.0465	-0.0112	0.0640
							(0.074)	(0.048)	(0.050)	(0.077)	(0.047)	(0.052)
Adj. R2	0.3587	0.4377	0.3135	0.3686	0.4428	0.3627	0.3289 to 0.3528	0.4138 to 0.4253	0.2606 to 0.2737	0.3581	0.4264	0.2762
Observations	1355	5414	6784	1295	5414	6779	1.334 to 1.355	5.388-5.414	5.388–5.414 6.767–6.784 1334	1334	5388	6767

Table 7 continued	þ											
	(13) Separated	(14)	(15)	(16) Joint	(17)	(18)	(19) Separated	(20)	(21)	(22) Joint	(23)	(24)
	Low educ.	Medium educ.	High educ.	Low educ.	Medium educ.	High educ.	Low educ.	Medium educ.	High educ.	Low educ.	Medium educ.	High educ.
PIAAC score in numeracy	0.0587***_ 0.0871*** /0.023	0.0847***_ 0.1028*** 0.0010	0.1446***_ 0.1578*** 0.0140	0.0589**	0.0909***	0.1527***	0.0554***- 0.0824***	0.0793***_ 0.0964***	0.0554*** 0.0793*** 0.1144*** 0.0553** 0.0824*** 0.0964*** 0.1255*** 0.0331 0.0131 0.0131		0.0856***	0.1220***
Years of schooling	(770.0)	(110.0)	(+10.0)	(070.0)	((10.0)	(+10.0)	0.0136-0.0158	0.0484***_ 0.0497***	 * * * *	0.0160*	0.0472***	0.0809***
							(0.012)	(0.010)	(0.007)	(600.0)	(0.010)	(0.007)
Non-cognitive skills test based measures	test based measu	ures										
Time per test item	-0.0044 (0.017)	0.0000 (0.002)	-0.0543*** (0.017)	0.0780 (0.092)	0.0003 (0.001)	- 0.0066 (0.009)	-0.0022 (0.014)	0.0005 (0.002)	$\begin{array}{r} -0.0470^{***} \ 0.1000 \\ (0.016) \ (0.094) \end{array}$	0.1000 (0.094)	0.0009 (0.001)	-0.0029 (0.009)
Time per correct test -0.0219 item (0.019)	-0.0219 (0.019)	-0.0262^{**} (0.010)	-0.0423 *** (0.011)	-0.0409 (0.043)	-0.0254** (0.011)	-0.0411^{***} (0.013)	-0.0209 (0.017)	-0.0256** (0.010)	-0.0379^{***} (0.010)	- 0.0469 (0.042)	-0.0249** (0.011)	-0.0386^{**} (0.012)
Skipped	-0.0210 (0.014)	-0.0195^{**} (0.008)	-0.0157 (0.013)	-0.0385^{**} (0.016)	-0.0250** (0.010)	-0.0243 (0.015)	-0.0208 (0.014)	-0.0186^{**} (0.008)	- 0.0119 (0.012)	-0.0388** (0.016)	-0.0229** (0.010)	-0.0186 (0.014)
Not attempted	0.02 <i>57</i> (0.020)	0.0068 (0.008)	0.0121 (0.010)	0.0373 (0.028)	0.0138 (0.009)	0.0157 (0.012)	0.0246 (0.020)	0.0064 (0.008)		0.0363 (0.028)	0.0125 (0.009)	0.0138 (0.012)

	(13) Separated	(14)	(15)	(16) Joint	(17)	(18)	(19) Separated	(20)	(21)	(22) Joint	(23)	(24)
	Low educ.	Medium educ.	High educ.	Low educ.	Medium educ.	High educ.	Low educ.	Medium educ.	High educ.	Low educ.	Medium educ.	High educ.
No. of actions/ test item	-0.0088 (0.012)	-0.0040 (0.008)	- 0.0078 (0.006)	- 0.0231 (0.019)	-0.0040 (0.013)	- 0.0038 (0.013)	-0.0132 (0.012)	-0.0039 (0.008)	- 0.0053 (0.006)	-0.0255 (0.019)	-0.0023 (0.013)	-0.0030 (0.013)
No. of	-0.0048	-0.0057	-0.0041	0.0128	0.0037	0.0074	-0.0113	-0.0065	-0.0019	0.0064	0.0014	0.0082
actions/correct test item	(0.013)	(0.008)	(0.004)	(0.019)	(0.013)	(0.010)	(0.013)	(0.008)	(0.004)	(0.019)	(0.013)	(0.010)
Non-cognitive skills measures from Background questionnaire	s measures from .	Background que	stionnaire									
Don't know	-0.0460	-0.0377*	-0.0412^{**}	-0.0545	-0.0358*	-0.0350*	-0.0417	-0.0337*	-0.0434^{**}	-0.0503	-0.0321	-0.0374*
answer	(0.041)	(0.021)	(0.021)	(0.038)	(0.021)	(0.021)	(0.043)	(0.020)	(0.019)	(0.040)	(0.020)	(0.020)
Refused to	-0.0230	-0.0055	0.0722	-0.0226	-0.0190	0.0717	-0.0173	-0.0062	0.0619	-0.0141	-0.0197	0.0621
answer	(0.070)	(0.047)	(0.050)	(0.073)	(0.046)	(0.050)	(0.073)	(0.046)	(0.048)	(0.075)	(0.045)	(0.048)
Adj. R2	0.3502 - 0.3634	0.4375 - 0.4392	0.3123 - 0.3181	0.3665	0.4411	0.3197	0.3605– 0.3729	0.4427– 0.4442	0.3612 - 0.3659	0.3761	0.4457	0.3672
Observations	1334-1355	5388-5414	6767–6784	1334	5388	6767	1274-1295	5388-5414	6774-6779	1274	5388	6762
Dependent variable: log gross hourly wages. Estimations using sample for female and country fixed effects Standard errors in parentheses * $p < 0.10, ** p < 0.05, *** p < 0.01$:: log gross hourl ntry fixed effects 2arentheses * p <	y wages. Estimat 	wages. Estimations using sample weights and 80 replicates for standard errors. All estimations include a constant, experience, the square of experience, a dummy $0.10, ** p < 0.05, *** p < 0.01$	e weights and 8	0 replicates for	standard erro	rs. All estimati	ons include a	constant, exper	ience, the sq	uare of experi	ence, a dummy

Table 7 continued

find very similar effects of non-cognitive skills on earnings among the three subgroups, but statistically significant only for the groups of low-educated and medium-educated individuals. The incidence of skipped questions in these two groups is also higher: 10 and 6% respectively, while in the group of high-educated workers is less than 3%. This reinforces the general result that we emphasized in Sect. 5: conditional on having the same level of cognitive test scores and the same number of years of schooling, individuals that do not answer questions in a non-penalized test have lower earnings, regardless the level of education they achieved.

The rate of "don't know" answers in the background questionnaire seems to have a smaller effect on earnings as the level of education increases. However, when we take into account the PIAAC proficiency level and the years of schooling, we find that this measure of non-cognitive skills affects only earnings of the high-educated workers.

In Table 3, we saw that time per correct answer had a negative and significant effect on earnings across all the specifications. Here we observe that when cognitive skills are not taken into account, this measure of non-cognitive skills has a statistically significant impact only in the group of high-educated workers. When PIAAC scores and years of schooling are controlled for, the effect is significant also for the medium-educated workers and, moreover, it increases slightly with the level of education: from 2.5% for medium-educated to 3.86% for high-educated.

Finally, for the rest of our non self-reported measures of non-cognitive skills we do not observe any significant pattern by level of education.

7.2 Differences by age group

In this heterogeneity analysis, we extend the sample to full-time 25–64 years employees and divide them in three groups: entry-age employees (25–34 years), prime-age employees (35–54 years) and exit-age employees (55–64 years).

When we look at the non-cognitive skills test based measures, we observe that "Time per correct test item" and the proportion of skipped questions in the test are significantly and negatively correlated with earning only in the case of prime-age and exit-age workers, when controlling for PIAAC scores and for the rest of non-cognitive skills measures as well (columns 16, 17 and 18 in Table 8). The magnitude of the impact is slightly bigger for older workers (55–64). However, when the level of education is accounted for, the coefficient is statistically significant only for the sample of prime-age workers.

The proportion of not attempted items in the test has a positive and statistically significant coefficient for all the age groups. An interesting result arises when we look at the number of actions per test item, whose coefficient is negative and statistically significant for the group of exit-age workers only, when controlling for both cognitive skills and years of schooling. Decreasing by one standard deviation the number of actions per test item, increases earnings between 4.4 and 4.8%, which is an economically important impact if we compare it to the increase in earnings of about 5.5% due to one more year of schooling in this group age.

In the case of non-cognitive skills measures based on the background questionnaire, it is worth noting that for the group of entry-age workers, the proportion of "Don't

Table 8 Non-cognitive skills in the Mincerian equation by groups of age (non-self reports)	nitive skills i	n the Mince	rian equatio	n by groups	of age (non	-self reports	()					
	(1)	(2)	(3)	(4)	(5)	(9)	(7) Separated	(8)	(6)	(10) Joint	(11)	(12)
	Entry-age	Prime-age	Exit-age	Entry-age	Prime-age	Exit-age	Entry-age	Prime-age	Exit-age	Entry-age	Prime-age	Exit-age
PIAAC score in numeracy	0.1536^{***} (0.008)	0.1873*** (0.007)	0.1978*** (0.019)	0.0934*** (0.008)	0.1032*** (0.008)	0.1324*** (0.021)						
Years of schooling				0.0486^{***}	0.0641^{***}	0.0556***						
				(0.003)	(0.002)	(0.005)						
Non-cognitive test based measures	sed measures											
Time per test item							-0.0025	-0.0029	0.0213	-0.0205	-0.0028^{**}	0.0244
							(0.022)	(0.002)	(0.069)	(0.071)	(0.001)	(0.053)
Time per correct test							-0.0046	-0.0079	-0.0196	-0.0033	-0.0115*	-0.0311
item							(0.010)	(0.007)	(0.015)	(0.031)	(0.007)	(0.023)
Skipped							-0.0769^{***}	-0.0900***	-0.0843^{***}	-0.0855^{***}	-0.1093^{***}	-0.1183^{***}
							(0.007)	(0.006)	(0.012)	(0.00)	(0.008)	(0.013)
Not attempted							-0.0220^{***}	-0.0208^{***}	-0.0016	0.0121	0.0168^{**}	0.0263^{***}
							(0.006)	(0.007)	(0.009)	(0.010)	(0.008)	(0.010)
No. of actions/test							0.0161^{*}	0.0072	-0.0262	-0.0183	-0.0306^{***}	-0.0732^{***}
item							(0.00)	(0.006)	(0.017)	(0.016)	(0.009)	(0.017)

446

Table 8 continued												
	(1)	(2)	(3)	(4)	(5)	(9)	(7) Separated	(8)	(6)	(10) Joint	(11)	(12)
	Entry-age	Prime-age	Exit-age	Entry-age	Prime-age	Exit-age	Entry-age	Prime-age	Exit-age	Entry-age	Prime-age	Exit-age
No. of actions/correct test item	est						0.0021 (0.008)	0.0003 (0.005)	-0.0148 (0.013)	0.0177 (0.014)	0.0211*** (0.008)	0.0396** (0.017)
Non-cognitive measures from Background questionnaire												
Don't know answer							-0.0625^{***}	-0.0873^{***}	-0.0673*	-0.0373^{***}	$-0.0373^{***} - 0.0770^{***} - 0.0271$	-0.0271
							(0.014)	(0.015)	(0.038)	(0.013)	(0.015)	(0.033)
Refused to answer							-0.0443	0.0548	-0.0806	-0.0316	0.0570*	-0.0326
							(0.066)	(0.034)	(0.086)	(0.055)	(0.034)	(0.074)
Adj. R2	0.3594	0.3605	0.3175	0.4209	0.4558	0.3844	0.278 to 0.3021	0.278 to 0.3021 0.2697 to 0.2997 0.2297 to 0.2578 0.3006	7 0.2297 to 0.2578	3 0.3006	0.3055	0.2797
Observations	8169	14,591	3437	8158	14,523	3408	8135-8169	14,518–14,591	3407–3437	8135	14,518	3407
PIAAC score in numeracy	0.1524^{***-} 0.1604^{***}		0.1770***- 0.1930***- 0.1599*** 0.1952*** 0.2134***		0.1838***	0.1980***	0.0932^{***} 0.0993^{***}	0.1024^{***-} 0.1098^{***}	0.1296^{***} . 0.1455^{***}	0.1009***	0.1058***	0.1355***
	(0.008)	(0.008)	(0.020)	(0.010)	(0.008)	(0.024)	(0.008)	(0.008)	(0.020) - (0.027) (0.010)	(0.010)	(6000)	(0.026)
Years of schooling							0.0481^{***} 0.0488^{***}	$0.0637^{***}_{0.0642^{***}}$	0.0546^{***} - 0.0562^{***}	0.0485***	0.0630***	0.0545***
							(0.003)	(0.002)	(0.006)	(0.003)	(0.002)	(0.005)

	(1)	(2)	(3)	(4)	(5)	(9)	(7) Separated	(8)	(6)	(10) Joint	(11)	(12)
	Entry-age	Prime-age	Exit-age	Entry-age	Entry-age Prime-age	Exit-age	Entry-age	Prime-age	Exit-age	Entry-age	Entry-age Prime-age	Exit-age
Non-cognitive skills test based measures												
Time per test item	$-0.0749^{***} - 0.0078$	-0.0078	0.0137	-0.0315	-0.0026	0.0212	$-0.0604^{***} - 0.0047$	-0.0047	0.0146	-0.0418	-0.0418 - 0.0004	0.0202
	(0.022)	(0.005)	(0.096)	(0.069)	(0.002)	(0.053)	(0.020)	(0.004)	(0.079)	(0.062)	(0.002)	(0.044)
Time per correct test	-0.0380^{***}	-0.0403^{***}	$-0.0403^{***} - 0.0453^{***}$	-0.0251	-0.0379***	-0.0490^{**}	-0.0379*** -0.0490** -0.0279***	$-0.0331^{***} - 0.0314^{**}$	-0.0314^{**}	-0.0101	-0.0321^{***}	-0.0318
Item	(0.009)	(0.007)	(0.016)	(0.030)	(0.008)	(0.024)	(0.008)	(0.006)	(0.015)	(0.027)	(0.007)	(0.021)
Skipped	-0.0049	-0.0175^{**}	-0.0071	-0.0120	-0.0309^{***}	-0.0384^{**} 0.0002	0.0002	-0.0000	0.0066	-0.0084	-0.0158^{**}	-0.0224
	(0.008)	(0.007)	(0.017)	(0.010)	(0.008)	(0.017)	(0.008)	(0.007)	(0.017)	(0.010)	(0.008)	(0.017)
Not attempted	0.0152^{**}	0.0222***	0.0339^{***}	0.0245*** 0.0289***	0.0289***	0.0402***	0.0131^{**}	0.0149*	0.0279^{**}	0.0216^{**}	0.0189*	0.0274^{**}
	(0.006)	(0.007)	(0.011)	(600.0)	(0.00)	(0.011)	(0.007)	(0.008)	(0.011)	(0.00)	(0.011)	(0.013)
No. of actions/test item	-0.0069	-0.0101^{**}	-0.0401^{***}	-0.0009	-0.007	$-0.0406^{**} - 0.0080$	-0.0080	-0.0084^{*}	-0.0444	-0.0085	-0.0040	-0.0485^{**}
	(0.008)	(0.005)	(0.013)	(0.016)	(0.008)	(0.017)	(0.008)	(0.005)	(0.013)	(0.014)	(0.008)	(0.017)
No. of actions/correct test item -0.0043	-0.0043	-0.0053	-0.0262^{**}	0.0060	0.0103	0.0163	-0.0023	-0.0058	-0.0259**	0.0111	0.0042	0.0172
	(0.007)	(0.004)	(0.013)	(0.013)	(0.007)	(0.016)	(0.007)	(0.004)	(0.012)	(0.012)	(0.007)	(0.016)

448

Table 8 continued

continued
×
e
abl
Ta

	(1)	(2)	(3)	(4)	(5)	(9)	(7) Separated	(8)	(6)	(10) Joint	(11)	(12)
	Entry-age	Prime-age	Exit-age	Entry-age	Prime-age	Exit-age	Entry-age	Prime-age	Exit-age	Entry-age	Prime-age	Exit-age
Non-cognitive skills measures from Background questionnaire												
Don't know answer	-0.0285^{**}	-0.0569^{***}	-0.0229	-0.0248*	-0.0541^{***}	-0.0058	0.0055	-0.0358^{**}	-0.0188	0.0000	-0.0337^{**}	-0.0080
	(0.013)	(0.014)	(0.037)	(0.013)	(0.014)	(0.035)	(0.013)	(0.014)	(0.038)	(0.013)	(0.014)	(0.036)
Refused to answer	-0.0448	0.0613*	-0.0833	-0.0433	0.0514	-0.0476	-0.0689*	0.0190	-0.1172*	-0.0689*	0.0126	-0.0861
	(0.042)	(0.033)	(0.073)	(0.040)	(0.033)	(0.070)	(0.040)	(0.029)	(0.065)	(0.038)	(0.029)	(0.061)
Adj. R2	0.3566– 0.3614	0.3595- 0.3636	0.3168– 0.3246	0.3621	0.3682	0.3378	0.4186– 0.4216	0.4548 - 0.4575	0.3839 - 0.3931	0.4225	0.4589	0.4001
Observations	8135-8169	14,518– 14,591	3407–3437	8135	14,518	3407	8124-8158	14,450– 14,523	3379–3408	8124	14,450	3379
Dependent variable: log gross hourly wages. Estimations using sample weights and 80 replicates for standard errors. All estimations include a constant, experience, the square of experience, a dummy	og gross hourly v	vages. Estimatio	ns using sample	e weights and 80	0 replicates for s	tandard erro	rs. All estimatic	ons include a co	nstant, experien	ce, the square	of experience,	a dummy

for female and country fixed efficets Standard errors in parentheses * p < 0.10, *** p < 0.05, *** p < 0.01 know" answers has a negative and statistically significant coefficient, when PIAAC scores in Numeracy are controlled for. However, this effect disappears when we include years of schooling as well, and the correlation between this variable and earnings remains significant only for prime-age workers.

As argued by Hanushek et al. (2015), the earnings gradient for cognitive skills is lower in early career spans and higher later. In contrast, we find that the earnings gradient for some of our non-cognitive skills does not vary considerably across age groups, and the correlation between non-cognitive skills and earnings is strongest for prime-age workers.

7.3 Differences by gender

Some differences by gender arise as well. Time per correct test item is correlated with earnings for both males and females. The increase in male earnings due to spending on standard deviation less time per correct test item is 4%, which doubles the increase in female earnings (2%). In the last specification when we control for both PIAAC scores and years of schooling this non-cognitive skills measure is correlated only with female earnings. The same applies to the proportion of not attempted questions in the test, it is only correlated with female earnings, when cognitive skills and years of schooling is accounted for.

The increase in earnings due to skipping less questions is very similar for males and females when we control for PIAAC score in numeracy (between 2 and 3.5%, due to a decrease of one standard deviation in the proportion of skipped questions). This result could be explained by the finding of Hernández and Hershaff (2014) that females and males do not skip questions at significantly different rates. However, when years of schooling are included and all non-cognitive skills measures are added jointly, the variable Skipped is only significant for males.

When looking at the measures based on the background questionnaire, the ratio of "Don't know" answers in the questionnaire is only significant for males (see Table 9, columns 14 and 16) when we control for both the score in numeracy and years of schooling.

In general, male earnings show a stronger correlation with non-cognitive skills than female earnings-

8 Robustness checks

8.1 Principal component analysis

In this section we implement Principal Component Analysis to reduce the number of variables representing non-cognitive skills. Our first observation is that the proposed measures of non-cognitive skills exhibit a low level of unidimensionality. The Principal Component analysis reveals that each of the fifth first components accounts from a 19.29 to a 9.33% of the total variance, thus it is quite disperse (see Table 10). The low unidimensionality is confirmed by the Cronbach alpha which equals 0.27. A potential explanation for this finding is that, in contrast to cognitive skills, personality traits

Table 9 Non-cognitive skills in	ve skills in the Minc	cerian equation by	the Mincerian equation by gender (non-self reports)	reports)				
	(1)	(2)	(3)	(4)	(5) Separated	(9)	(7) Joint	(8)
	Female	Male	Female	Male	Female	Male	Female	Male
PIAAC score in numeracy	0.1906^{***} (0.010)	0.1816^{***} (0.010)	0.1050^{***} (0.011)	0.1046^{***} (0.010)				
Years of schooling			0.0680***	0.0588^{***}				
			(0.003)	(0.003)				
Non-cognitive skills test based measures	st based measures							
Time per test item					0.0000	-0.0259*	-0.0036^{***}	-0.0246
					(0.001)	(0.015)	(0.001)	(0.050)
Time per correct					0.0095	-0.0166*	0.0056	-0.0098
test item					(0.010)	(0.009)	(0.010)	(0.020)
Skipped					-0.0818^{***}	-0.0978^{***}	-0.1024^{***}	-0.1167^{***}
					(0.008)	(0.00)	(0000)	(0.011)
Not attempted					-0.0135	-0.0237 **	0.0254^{***}	0.0121
					(0.008)	(0.010)	(00.0)	(0.013)
No. of actions/test					0.0131*	0.0003	-0.0137	-0.0489^{***}
item					(0.007)	(0.009)	(0.011)	(0.015)
No. of					0.0033	-0.0047	0.0103	0.0325^{**}
actions/correct test item					(0.005)	(0.009)	(600.0)	(0.014)

 ${\textcircled{\sc D}} Springer$

Table 9 continued	þ							
	(1)	(2)	(3)	(4)	(5) Separated	(9)	(7) Joint	(8)
	Female	Male	Female	Male	Female	Male	Female	Male
Non-cognitive skil	Non-cognitive skills measures from Background questionnaire	ound questionnaire						
Don't know					-0.0713^{***}	-0.1017^{***}	-0.0638^{***}	-0.0894^{***}
answer					(0.022)	(0.022)	(0.022)	(0.021)
Refused to answer					0.0884^{**}	0.0490	0.0993^{**}	0.0405
					(0.042)	(0.044)	(0.045)	(0.044)
Adj. R2	0.3952	0.2852	0.5005	0.3736	0.3031-0.335	0.1874-0.2214	0.3424	0.2279
Observations	6736	7855	6708	7815	6699-6736	7819–7855	6699	7819
PIAAC score in	0.1893***-0.1981***	1981*** 0.1709***-0.1906***	0.1841^{***}	0.1776^{***}	**_0.1096***	0.1030***-0.1127***	0.1082^{***}	0.1052^{***}
numeracy	(0.011)	(0.010)	(0.011)	(0.012)	(0.011)	(0.010) - (0.012)	(0.012)	(0.012)
Years of schooling	•				0.0676***-0.0681***	0.0676***-0.0681*** 0.0583***-0.0589*** 0.0670***	0.0670^{***}	0.0578***
					(0.003)	(0.003)	(0.003)	(0.003)
Non-cognitive skil	Non-cognitive skills test based measures							
Time per test item -0.0012	-0.0012	-0.0589	-0.0033^{**}	-0.0277	-0.0003	-0.0430	-0.0008	0.0017
	(0.002)	(0.036)	(0.002)	(0.057)	(0.001)	(0.034)	(0.001)	(0.060)
Time per correct	-0.0249^{***}	-0.0475^{***}	-0.0238^{**}	-0.0323	-0.0201^{***}	-0.0399^{***}	-0.0221^{***}	-0.0368
test item	(0.00)	(0.011)	(0.009)	(0.026)	(0.007)	(600.0)	(0.008)	(0.025)
Skipped	-0.0171^{**}	-0.0200*	-0.0302^{***}	$-0.0302^{***} - 0.0355^{***} - 0.0043$	-0.0043	-0.0133	-0.0077	-0.0244^{**}

	(1)	(2)	(3)	(4)	(5) Separated	(9)	(7) Joint	(8)
	Female	Male	Female	Male	Female	Male	Female	Male
	(0.007)	(0.011)	(0.009)	(0.012)	(0.007)	(0.010)	(0.008)	(0.012)
Not attempted	0.0277^{***}	0.0191^{*}	0.0347***	0.0258*	0.0156^{**}	0.0151	0.0179^{**}	0.0200
	(0.008)	(0.010)	(0.008)	(0.013)	(0.007)	(0.011)	(0.008)	(0.016)
No. of actions/test	-0.0004	-0.0215^{**}	-0.0000	-0.0195	0.0012	-0.0205 **	0.0081	-0.0191
item	(0.006)	(6000)	(0.011)	(0.015)	(0.005)	(0.008)	(0.010)	(0.014)
No. of actions/correct test item	0.0013	-0.0171^{**}	0.0062	0.0117	0.0005	-0.0175^{**}	-0.0007	0.0088
	(0.005)	(600.0)	(0.009)	(0.014)	(0.004)	(0.008)	(0.007)	(0.014)
Non-cognitive skills measures from	om Background questionnaire	stionnaire						
Don't know	-0.0512^{**}	-0.0641^{***}	-0.0483^{**}	-0.0617^{***}	-0.0210	-0.0502**	-0.0197	-0.0487^{**}
answer	(0.020)	(0.022)	(0.021)	(0.021)	(0.020)	(0.021)	(0.020)	(0.020)
Refused to answer	0.0877*	0.0570	0.0808*	0.0418	0.0572*	0.0057	0.0531	-0.0052
	(0.045)	(0.042)	(0.045)	(0.042)	(0.034)	(0.039)	(0.034)	(0.040)
Adj. R2	0.3951 - 0.3975	0.2831 - 0.2892	0.4031	0.2942	0.5003-0.5012	0.3719-0.376	0.5023	0.3786
Observations	6699–6736	7819–7855	6699	7819	6671–6708	7779–7815	6671	<i>9777</i>
Dependent variable: log gross hourly wages. Estimations usit of experience, a dummy for female and country fixed effects Standard errors in parentheses $* p < 0.10, ** p < 0.05, ***$	urly wages. Estimati ale and country fixed p < 0.10, ** p < 0	ly wages. Estimations using sample weights and 80 replicates for standard errors. All estimations include a constant, experience, the square e and country fixed effects $< 0.10, ** p < 0.05, *** p < 0.01$	eights and 80 rep	olicates for standa	rd errors. All estima	tions include a con:	stant, experien	ce, the square

SERIEs (2017) 8:417-473

Table 9 continued

0.8718

0.9277

0

rubie ro Trincipal components			
Ppal componen\Var.	Eigenvalue	Var. proportion	Cumulative
1st Component	2.12168	0.1929	0.1929
2nd Component	1.4531	0.1321	0.325
3rd Component	1.25563	0.1141	0.4391
4th Component	1.13042	0.1028	0.5419
5th Component	1.02607	0.0933	0.6352
6th Component	0.959718	0.0872	0.7224
7th Component	0.930084	0.0846	0.807

0.0649

0.0559

0.0532

0.0191

0.713482

0.614928

0.584808

0.210092

Table 10 Principal components

have been found to be multidimensional rather than unitary (Almlund et al. 2011). The most prominent taxonomy is the Big-Five (see McCrae and John 1992), which classifies personality traits in five independent categories (agreeableness, conscientiousness, extraversion, openness to experience and neuroticism). The fact that the proposed measures do not a capture a unidimensional variable may be explained by the multidimensionality in the variables of interest. Moreover, as previously argued, some of our test based measures might be capturing cognitive skills.

Applying the Kaiser (1960) criterion to reduce the number of variables in the analysis, we retain the first five components.¹⁸ In Table 11 we report the loadings of each original variable in the retained components (to make easier its interpretation we highlighted in bold values above 0.3). As we can see the first and the fourth components seem to be related with test-based measures, despite not offering a very clear pattern of a latent structure. The second component presents its higher correlations with self-reported measures of non-cognitive skills. The third component captures a mix of time per question and self-reported measures presenting its higher correlation with time per question. Finally, the fifth component is mainly capturing non-self-reported measures based on performance in the background questionnaire ("don't know" and refused items).

We replicate our main regressions after substituting the set of proposed variables for non-cognitive skills by the first five components that survived the Kaiser (1960) criterion, see Table 12. When not controlling for cognitive skills (column 1), all factors are significantly associated to earnings except the fourth component. When including numeracy scores as a measure for cognitive skills (column 2), the first component is no longer statistically significant, while the third component also loses its significance after adding years of schooling as a control (column 3). Thus, the second component (associated to self-reported measures on political efficacy and social trust) and the fifth

8th Component

9th Component

10th Component

11th Component

 $^{^{18}}$ The Kaiser (1960) criterion proposes to retain only factors with eigenvalues greater than 1. It is one of the most extended criteria in this type of analysis.

455

Variable\Ppal component	1st Component	2nd Component	3rd Component	4th Component	5th Component
Time per test item	0.2506	0.2318	0.4572	0.4713	0.0404
Time per correct test item	0.346	0.2458	0.4074	0.3271	0.0078
Skipped -	- 0.3193	0.4165	0.2781	0.3038	0.1272
Not attempted -	- 0.2747	0.3433	0.3758	0.3918	0.1566
No. of actions/ test item	0.5839	0.1261	0.2065	0.2695	0.032
No. of actions/ correct test item	0.5146	0.2736	0.0929	0.4321	0.09
Don't know - answer	- 0.0306	0.1433	0.0402	0.118	0.6355
Refused to - answer	- 0.0216	0.055	0.0301	0.1323	0.7324
Cultural engagement	0.0861	0.2588	0.2866	0.0588	0.0791
Political efficacy	0.1195	0.4623	0.3373	0.2594	0.0116
Social trust	0.1055	0.4467	0.3958	0.2521	0.0416

Table 11 Correlation between principal components and variables for non-cognitive skills

component (associated to "don't know" and refused items) are the ones that present a robust correlation with individual earnings. This is consistent with our previous results in which "don't know" answers and self-reports on social trust and political efficacy were the variables presenting a more robust correlation with earnings. Also note that, the first component which is the one more clearly related to test based measures, is the one that clearly drops its significance after including numeracy scores.

8.2 Additional controls

For our main estimations we followed the specification of the earnings equation proposed by Hanushek et al. (2015) which uniquely controls for years of schooling, gender, the second degree polynomial of experience and country fixed effects. There is a wide range of elements that potentially affect individual earnings and that are not contemplated among the previous regressors. Consequently, in this section we include additional controls available in the PIAAC database.¹⁹ In particular, we include health status (in three categories: excellent-very good, good-fair or poor), type of occupation (in four categories: high skill occupations, medium-skill white collar occupations, medium-skills blue collar occupations and low skill occupations), economic sector

¹⁹ Despite extending the number of controls, it is difficult to completely exclude the possibility of an omitted variables bias when analyzing cross sectional data as PIAAC. For instance, data on disabilities or ethnicity is not available in the PIAAC database.

	(1)	(2)	(3)
PIAAC score in numeracy		0.1798***	0.1045***
		(0.009)	(0.009)
Years of schooling			0.0618***
			(0.002)
1st Principal component	0.0310***	-0.0005	-0.0046
	(0.004)	(0.005)	(0.004)
2nd Principal component	-0.0871***	-0.0469^{***}	-0.0267***
	(0.004)	(0.005)	(0.005)
3rd Principal component	0.0155***	0.0187***	0.0077
	(0.005)	(0.006)	(0.006)
4th Principal component	-0.0086	0.0225***	0.0136*
	(0.007)	(0.008)	(0.007)
5th Principal component	-0.0572^{***}	-0.0500***	-0.0366^{**}
	(0.016)	(0.016)	(0.015)
Adj. R2	0.31	0.37	0.46
Observations	14,471	14,471	14,404

Table 12 Non-cognitive skills in the Mincerian equation by level of education using principal components

Dependent variable: log gross hourly wages

Standard errors in parentheses * p < 0.10, ** p < 0.05, *** p < 0.01

(private vs public-non for profit) and immigration status (first or second generation immigrant vs not immigrant).

Results are available in Tables 13 and 14. Our results are qualitatively similar to the ones in our main specifications in Tables 3 and 4. "Don't know" answers, "time per correct test item" and self-reports on social trust remain statistically significant under all specifications. In contrast, some of the variables that were presenting a less robust association with earnings, like "skipped test items", "non-attempted test items" and background questionnaire "refusals" now exhibit an even weaker association. Self-reports on "political efficacy" lose some significance after adding controls but still remain significant in most specifications. In summary, by including these additional controls we confirm our results for the variables that exhibited a more robust association with earnings in our main analysis.

The specification proposed by Hanushek et al. (2015) controls for a quadratic polynomial in actual work experience. Experience measured this way could be endogenous to skills and it could be one of the channels through which skills could affect individual earnings. In a robustness check, we replace this measure by potential experience, measured as age minus years of schooling minus six. Results do not change considerably with respect to our main specification.²⁰

²⁰ They are available upon request.

	(1)	(2)	(3) Separated	(4) Joint	(5) Separated	(6) Joint	(7) Separated	(8) Joint
PIAAC score in numeracy	0.1211^{***}	0.0845***			0.1157***-0.1272***	0.1235***	0.0809 * * * - 0.0904 * * *	0.0888***
	(0.008)	(0.008)			(0.008)	(0.00)	(0.008)	(0.010)
Years of schooling		0.0417^{***}					$0.0416^{***}-0.0418^{***}$	0.0413^{***}
		(0.003)					(0.003)	(0.003)
Non-cognitive skills test based measures	neasures							
Time per test item	0.0007	0.0013	-0.0028	0.0011	-0.0023	0.0015	0.0007	0.0013
	(0.002)	(0.001)	(0.003)	(0.002)	(0.003)	(0.001)	(0.002)	(0.001)
Time per correct test item	-0.0070	-0.0117*	-0.0290^{***}	-0.0304^{**}	-0.0284^{***}	-0.0294^{***}	-0.0070	-0.0117*
	(0.006)	(0.006)	(0.006)	(0.007)	(0.006)	(0.006)	(0.006)	(0.006)
Skipped	-0.0517^{***}	-0.0593^{***}	-0.0098	-0.0134^{*}	- 0.0066	-0.0086	-0.0517^{***}	-0.0593^{***}
	(0.006)	(0.007)	(0.007)	(0.008)	(0.007)	(0.008)	(0.006)	(0.007)
Not attempted	-0.0109*	0.0076	0.0146^{**}	0.0165^{**}	0.0122*	0.0131	-0.0109*	0.0076
	(0.006)	(0.008)	(0.006)	(0.008)	(0.007)	(0.00)	(0.006)	(0.008)
No. of actions/test item	0.0063	-0.0119	-0.0037	-0.0002	-0.0051	0.0008	0.0063	-0.0119
	(0.005)	(0.008)	(0.005)	(0.008)	(0.004)	(0.008)	(0.005)	(0.008)
No. of actions/correct test item	0.0012	0.0108*	-0.0020	0.0048	-0.0041	0.0017	0.0012	0.0108*
	(0.004)	(0.006)	(0.004)	(0.006)	(0.003)	(0.006)	(0.004)	(0.006)

 Table 13
 Non-cognitive skills in the Mincerian equation (non-self reports). Additional controls

continued	
13	
Table	

D Springer

	(1)	(2)	Separated	Joint	Separated	Joint	Separated	Joint
Non-cognitive skill measures from	Bac	kground questionnaire	٩					
Don't know answer	-0.0477 ***	-0.0446^{***}	-0.0338^{**}	-0.0326^{**}	-0.0262*	-0.0251*	-0.0477***	-0.0446^{***}
	(0.016)	(0.016)	(0.015)	(0.015)	(0.015)	(0.015)	(0.016)	(0.016)
Refused to answer	0.0293	0.0319	0.0370	0.0313	0.0214	0.0166	0.0293	0.0319
	(0.026)	(0.027)	(0.027)	(0.027)	(0.026)	(0.027)	(0.026)	(0.027)
Adj. R2	0.47	0.5	0.44-0.45	0.45	0.47	0.47	0.5	0.5
No. of observations 13,613	13,613	13,551	13,485–13,551	13,547	13,547–13,613	13,547	13,485–13,551	13,485

or experience, a dummy for remain and country fixed effects Standard errors in parentheses * p < 0.10, ** p < 0.05, *** p < 0.01

Table 14 Non-cognitive skills in the Mincerian equation (self reports). Additional controls	1 the Mincerian equ	ation (self reports). Ad	ditional controls			
	(1) Separated	(2) Joint	(3) Separated	(4) Joint	(5) Separated	(6) Joint
PIAAC score in numeracy			0.1192***_ (0.008)	0.1216*** (0.009)	0.0837***_0.0849*** (0.008)	0.0881***
Years of schooling					(0.001) 0.0811***-0.0849*** 0.003	0.0408***
Self-reported non-cognitive					(cnn.n)	(con.n)
Cultural engagement	0.0016	-0.0035	-0.0052	-0.0071	-0.0063	-0.0076
	(0.005)	(0.005)	(0.005)	(0.005)	(0.005)	(0.005)
Political efficacy	0.0216^{***}	0.0141^{**}	0.0139^{**}	0.0088	0.0095*	0.0066
	(0.006)	(0.006)	(0.005)	(0.006)	(0.005)	(0.006)
Social trust	0.0298^{***}	0.0244^{***}	0.0244^{***}	0.0223^{***}	0.0158***	0.0145^{**}
	(0.006)	(0.007)	(0.006)	(0.006)	(0.006)	(0.006)
Non self-reported non-cognitive skills	skills					
Time per test item		0.0014		0.0012		0.0015
		(0.001)		(0.002)		(0.001)
Time per correct test item		-0.0111*		-0.0297^{***}		-0.0288^{***}
		(0.006)		(0.007)		(0.006)
Skipped		-0.0574^{***}		-0.0129		-0.0086
		(0.007)		(0.008)		(0.008)
Not attempted		0.0072		0.0161^{**}		0.0131
		(0.008)		(0.008)		(0.009)
No. of actions/test item		-0.0117		-0.0003		0.0007
		(0.008)		(0.008)		(0.008)

	(1) Separated	(2) Joint	(3) Separated	(4) Joint	(5) Separated	(6) Joint
No. of actions/correct test item		0.0103		0.0045		0.0015
		(0.006)		(0.006)		(0.006)
Don't know answer		-0.0490^{***}		-0.0365^{**}		-0.0283*
		(0.017)		(0.016)		(0.016)
Refused to answer		0.0378		0.0366		0.0233
		(0.029)		(0.029)		(0.028)
Adj. R2	0.44	0.45	0.47	0.47	0.5	0.5
Observations	13,576-13,610	3505	13,576–13,610	13,505	13,542-13,515	13,444

of experience, a dummy for female and country fixed effects Standard errors in parentheses * p < 0.10, *** p < 0.05, *** p < 0.01

460

Table 14 continued

8.3 Additional measures of non-cognitive skills

An alternative measure of non-cognitive skills is constructed using a set of self-reported variables on learning strategies, as in Cabrales et al. (2014). This could be considered as a measure of the degree of motivation. We create a dummy variable which takes the value 1 if the person responds that she feels identified "To some extent" or "To a high extent" to at least one of the following questions: Relate new ideas into real life, Like learning new strategies, Attribute something new, Get to the bottom of difficult things, Figure out how different ideas fit together and Looking for additional info. Then, we standardize this dummy with respect to the mean and standard deviation of the pool of countries (as for the rest of the variables).²¹ Results are available in the "Online Appendix". The proxy for motivation is strongly and positively correlated with earnings in all specifications. Moreover, its sign and statistical significance do not change when we include it jointly with the other self-reported and non self-reported measures of non-cognitive skills. Overall, results for the rest of the measures hardly change with respect to those reported in Table 4.

8.4 Extended sample including all workers aged 25–64

In order to work with a homogenous sample, as in Hanushek et al. (2015), we restricted our sample to prime age workers (35–54 years). To show the relevance of this sample restriction, in Tables 15 and 16 we estimate using the extended sample of employees: 25–64 years, which includes entry-age (25–34 years), prime-age (34–54 years) and exit-age workers (55–64 years).

The main conclusions that we draw from Tables 3 and 4 maintain when we extend the sample. Time per correct test item and Skipped questions show the same stable coefficients across all the specifications and, moreover, the magnitude of the coefficients is very similar to that estimated for the prime-age sample (almost 3 and 1.6% increase in earnings). The number of actions per test item is now strongly significant in the last specification where we control for both the PIAAC score and the number of years of schooling and include jointly all the non-cognitive skills measures (before it was significant only when the non-cognitive skills measures were included separately). The ratio of "Don't know" answers in the questionnaire is still negatively correlated with individual earnings, even though it is not statistically significant in the last specification when we include all the measures of non-cognitive skills jointly. One important difference with respect to the restricted sample, is that now the ratio of "Refused to answer" questions in the questionnaire is also negatively correlated and statistically significant when we control for both years of schooling and cognitive.

Finally, coefficients of self-reported measures of non-cognitive skills in the extended sample are very similar to the ones from Table 4, though they are slightly smaller in magnitude.

²¹ We could also calculate the average of these variables and then standardize it with respect to the mean and standard deviation of the pool of countries. Results are the same and they are available upon request.

	(1)	(2)	(3) Separated	(4) Joint	(5) Separated	(6) Joint	(7) Separated	(8) Joint
PIAAC score in numeracy	0.1821*** (0.006)	0.1033^{***} (0.006)			0.1742***_0.1893*** (0.006)	0.1813*** (0.007)	0.1001***_0.1089*** (0.006)	0.1066*** (0.007)
Years of schooling		0.0591***					0.0587***_0.0592***	0.0583*** (0.002)
Non-cognitive skills test based measures	measures	~					~	~
Time per test item			0.0058	0.0068	-0.0024	0.0058	-0.0002	0.0062
			(0.010)	(0.012)	(0.011)	(0.011)	(0.09)	(0.009)
Time per correct test item			-0.0054	-0.0124*	-0.0373^{***}	-0.0376^{***}	-0.0289^{***}	-0.0291^{***}
			(0.006)	(0.007)	(0.006)	(0.007)	(0.005)	(0.006)
Skipped			-0.0884^{***}	-0.1072^{***}	-0.0137^{**}	-0.0279^{***}	-0.0059	-0.0160^{***}
			(0.005)	(0.006)	(0.006)	(0.006)	(0.005)	(0.006)
Not attempted			-0.0195^{***}	0.0174^{***}	0.0224^{***}	0.0303^{***}	0.0165^{***}	0.0208^{***}
			(0.005)	(0.006)	(0.005)	(0.006)	(0.005)	(0.007)
No. of actions/test item			0.0034	-0.0374^{***}	-0.0147^{***}	-0.0155^{**}	-0.0147^{***}	-0.0170^{**}
			(0.005)	(0.008)	(0.004)	(0.007)	(0.004)	(0.008)
No. of actions/correct test item	_		-0.0029	0.0238^{***}	-0.0088^{**}	0.0110*	-0.0084^{**}	0.0105
			(0.004)	(0.007)	(0.004)	(0.006)	(0.003)	(0.006)

Table 15Non-cognitive skills in the Mincerian equation (non-self reports)

continue
15
Table

	(1)	(2)	(3) Separated	(4) Joint	(5) Separated	(6) Joint	(7) Separated	(8) Joint
Non-cognitive skills measures from Background questionnaire	easures from E	3ackground qu	estionnaire					
Don't know answer			-0.0777 * * *	-0.0587^{***}	-0.0437^{***}	-0.0389^{***}	-0.0204*	-0.0170
			(0.012)	(0.012)	(0.012)	(0.011)	(0.012)	(0.012)
Refused to answer			0.0039	0.0125	0.0064	0.0033	-0.0318*	-0.0344^{**}
			(0.022)	(0.021)	(0.019)	(0.019)	(0.018)	(0.017)
Adj. R2	0.3833	0.465	0.299 - 0.3263	0.3284	0.3819-0.3851	0.3891	0.4638 - 0.4659	0.4677
No. of observations	26,128	26,020	25,885-26,020	25,992	25,992–26,128	25,992	25,885–26,020	25,885
Sample extended to employees 25-64 years old. Dependent variable: log gross hourly wages. Estimations using sample weights and 80 replicates for standard errors. All	nployees 25-6	4 years old. E	bependent variable: log	g gross hourly wage	s. Estimations using s	sample weights and	80 replicates for stand	ard errors. All

à 4 estimations include a constant, experience, the square of experience, a dummy for female and country fixed effects Standard errors in parentheses * p < 0.10, ** p < 0.05, *** p < 0.01

Table 16 Non-cognitive skills in the Mincerian equation (self reports)	s in the Mincerian e	quation (self reports)				
	(1) Separated	(2) Joint	(3) Separated	(4) Joint	(5) Separated	(6) Joint
PIAAC score in numeracy			0.1756***-0.1810***	0.1717***	0.1016***–0.1033***	0.1038***
Years of schooling			(0.000)	(0.007)	(0.000) 0.0577***_0.0591***	(0.007) 0.0565*** (0.003)
Self-reported non-cognitive skills measures	ills measures				(700.0)	(700.0)
Cultural engagement	0.0260^{***}	0.0118^{***}	0.0106^{**}	0.0048	-0.0000	-0.0028
	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)
Political efficacy	0.0560^{***}	0.0362^{***}	0.0370^{***}	0.0260^{***}	0.0192***	0.0139^{***}
	(0.005)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)
Social trust	0.0640^{***}	0.0462^{***}	0.0467^{***}	0.0385***	0.0254***	0.0220^{***}
	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)
Non self-reported non-cognitive skills measures	ve skills measures					
Time per test item		0.0068		0.0059		0.0062
		(0.011)		(0.010)		(0.00)
Time per correct test item		-0.0120*		-0.0359^{***}		-0.0283^{***}
		(0.007)		(0.007)		(0.006)
Skipped		-0.0995^{***}		-0.0264^{***}		-0.0163^{***}
		(0.005)		(0.006)		(0.006)
Not attempted		0.0182^{***}		0.0302^{***}		0.0213^{***}
		(0.006)		(0.006)		(0.007)

continue	
16	
Table	

	(1) Separated	(2) Joint	(3) Separated	(4) Joint	(5) Separated	(6) Joint
No. of actions/test item		-0.0357 ***		-0.0154^{**}		-0.0170^{**}
		(0.008)		(0.007)		(0.008)
No. of actions/correct test item		0.0221^{***}		0.0105^{*}		0.0102
		(0.007)		(0.006)		(0.006)
Don't know answer		-0.0638^{***}		-0.0448^{***}		-0.0240^{**}
		(0.012)		(0.012)		(0.012)
Refused to answer		0.0172		0.0075		-0.0295
		(0.022)		(0.020)		(0.019)
Adj. R2	0.2992-0.3121	0.3457	0.3836-0.3912	0.3991	0.4649-0.4675	0.4705
Observations	26,046-26,124	25,903	26,046-26,124	25,903	25,939–26,016	25,797
Sample extended to employees 25–64 years old. Dependent variable: log gross hourly wages. Estimations using sample weights and 80 replicates for standard errors. All estimations include a constant, experience, the square of experience, a dummy for female and country fixed effects Standard errors in parentheses * $p < 0.10$, ** $p < 0.05$, *** $p < 0.01$	-64 years old. Dependent variable perience, the square of experience, < 0.10, **p < 0.05, ***p < 0.01	variable: log gross ho erience, a dummy for f $\gamma < 0.01$	urly wages. Estimations emale and country fixed	using sample weights effects	and 80 replicates for star	ndard errors. All

SERIEs (2017) 8:417-473

8.5 Sample size across countries

As indicated in Table 1, samples of participating countries have different size. Given the pooled nature of our study, countries with larger number of observations may be contributing more to the obtained results. To study this possibility we re-estimate Eq. (2) separately for each country and then we average the estimates across countries. The results are reported in Tables 17 and 18. Results are qualitatively unchanged with respect the pooled sample with the only exception that "don't know" answers lose significance when jointly controlling for numeracy and years of schooling.

8.6 Alternative measures of cognitive skills

Following Hanushek et al. (2015) we have considered PIAAC numeracy scores as our baseline measure for non-cognitive skills. Our results do not change if we substitute numeracy scores by scores in the other two domains: literacy and problem solving. This results are available in the "Online Appendix".

9 Conclusions

One of the main contributions of the PIAAC database and similar international assessments is the provision of accurate measures of cognitive skills. However, international assessments usually do not contain information on a set of skills that has been identified to be as relevant as cognitive skills for individual success, namely non-cognitive skills. In the present paper, we complement the PIAAC database by constructing several measures of non-cognitive skills. An important advantage of our approach is that it provides a cheap and readily available way of adding information on non-cognitive skills.

Consistent with the previous literature, several of our proxies for non-cognitive skills have explanatory power on individual earnings beyond the one captured by PIAAC test scores. Some evidence arises from questionnaire non-response. First, because this variable has been previously validated as a proxy for non-cognitive skills related to the personality trait of conscientiousness and openness to experience (Hedengren and Stratmann 2012; Hitt et al. 2016; Cheng et al. 2016). Second, because this variable exhibits a robust correlation with individual earnings even after including other measures of human capital such as years of schooling and PIAAC test scores. Consistently with the previous literature, individuals with a higher incidence of non-response have a lower level of individual earnings.

We also exploit the computer based nature of the PIAAC test to construct other variables that can be potentially related to individuals' personality. Among the proposed variables, the ones that seem to provide relevant information on individual skills are skipped test item and time per correct test item. Despite these results need to be interpreted with caution, due to the correlation between these two test based measures and PIAAC scores, the fact that skipped test item enters in a statistically significant way in the earnings equation even after controlling for PIAAC test scores, suggests that they may be better capturing some non-cognitive skills than PIAAC test scores.

	(1)	(2)	(3) Separated	(4) Joint	(5) Separated	(6) Joint	(7) Separated	(8) Joint
PIAAC score in numeracy	0.177 *** (0.005)	0.101*** (0.005)			0.1622***_0.1822*** (0.005-0.006)	0.1648^{***} (0.006)	0.0924***-0.1055*** (0.005-0.006)	0.0971*** (0.006)
Years of schooling	~	0.0604***					0.0599***_0.0605***	0.0589*** (0.002)
Non-cognitive skills test based measures	l measures	~					~	~
Time per test item			0.0024	-0.0232	-0.0504^{***}	-0.0007	-0.0438***	0.0060
			(0.011)	(0.027)	(0.011)	(0.027)	(0.010)	(0.024)
Time per correct test item			-0.0010	0.0055	-0.0311^{***}	-0.0289^{**}	-0.0274^{***}	-0.0289^{**}
			(0.005)	(0.012)	(0.005)	(0.012)	(0.004)	(0.011)
Skipped			-0.0982^{***}	-0.111^{***}	-0.0290^{***}	-0.0385^{***}	-0.0175^{***}	-0.0217^{***}
			(0.005)	(0.005)	(0.005)	(0.006)	(0.005)	(0.005)
Not attempted			-0.0306^{***}	0.0148^{**}	0.0100	0.0218^{***}	0.0050	0.0122**
			(0.007)	(0.006)	(0.007)	(0.006)	(0.006)	(0.006)
No. of actions/test item			0.0060	-0.238^{***}	-0.0101^{***}	-0.0076	-0.0093^{***}	-0.0059
			(0.004)	(0.006)	(0.004)	(0.006)	(0.004)	(0.006)
No. of actions/correct test item	-		-0.0024	0.0113*	-0.0082^{**}	0.0034	-0.0072^{**}	0.0034
			(0.004)	(0.006)	(0.004)	(0.006)	(0.003)	(0.006)

Table 17Non-cognitive skills in the Mincerian equation (non-self reports)

	(1) (5	2) (3) Separated	(4) Joint	(5) Separated	(6) Joint	(7) Separated	(8) Joint
Non-cognitive skills measures from Background questionnaire	s from Backgro	und questionnaire					
Don't know answer		-0.0669^{***}	-0.0486^{***}	-0.0330^{**}	-0.0260*	-0.0225	-0.0178
		(0.014)	(0.014)	(0.014)	(0.014)	(0.014)	(0.013)
Refused to answer		0.0833	0.090*	0.0975**	0.0775**	0.0756**	0.0540
		(0.055)	(0.051)	(0.041)	(0.039)	(0.038)	(0.035)

Separate estimations for each country (the table reports average results with same weight for each country). Dependent variable: log gross hourly wages. Separate estimations for each country using sample weights and 80 replicates for standard errors. Average results with same weight for each country. All estimations include a constant, experience, the square of experience, a dummy for female and country fixed effects Standard errors in parentheses * p < 0.10, **p < 0.05, ***p < 0.01

468

Table 17 continued

Table 18 Non-cognitive skills in	s in the Mincerian e	the Mincerian equation (self reports)				
	(1) Separated	(2) Joint	(3) Separated	(4) Joint	(5) Separated	(6) Joint
PIAAC score in numeracy			0.1683***_0.1758*** (0.005_0.006)	0.1545*** (0.006)	0.0976–0.1005*** (0.005)	0.0942***
Years of schooling					0.0588***_0.0603***	0.0566***
Self-reported non-cognitive skills measures	ills measures					(200.0)
Cultural engagement	0.0261^{***}	0.0097**	0.0124^{***}	0.0040	0.0009	-0.0037
	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)
Political efficacy	0.0699***	0.0461^{***}	0.0478***	0.0356***	0.0273***	0.0212^{***}
	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)
Social trust	0.0645***	0.0408^{***}	0.0453***	0.0329^{***}	0.0250^{***}	0.0180^{***}
	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)
Non self-reported non-cognitive skills measures	ve skills measures					
Time per test item		-0.0146		0.0040		0.0097
		(0.027)		(0.027)		(0.025)
Time per correct test item		0.0019		-0.0292^{**}		-0.0294^{***}
		(0.012)		(0.012)		(0.011)
Skipped		-0.0985^{***}		-0.0334^{***}		-0.0199^{***}

SERIEs (2017) 8:417-473

	(1) Separated	(2) Joint	(3) Separated	(4) Joint	(5) Separated	(6) Joint
		(0.005)		(0.006)		(0.005)
Not attempted		0.0147^{**}		0.0211^{***}		0.0121*
		(0.006)		(0.006)		(0.006)
No. of actions/test item		-0.0207 ***		-0.0065		-0.0057
		(0.006)		(0.006)		(0.006)
No. of actions/correct test item		0.0093		0.0024		0.0026
		(0.006)		(0.006)		(0.005)
Don't know answer		-0.0487^{***}		-0.0304^{**}		-0.0216
		(0.014)		(0.014)		(0.014)
Refused to answer		0.1133^{***}		0.0953^{***}		0.0705^{**}
		(0.029)		(0.029)		(0.033)

sample weights and 80 replicates for standard errors. All estimations include a constant, experience, the square of experience, a dummy for female and country fixed effects Standard errors in parentheses * p < 0.10, ** p < 0.05, *** p < 0.01

Table 18 continued

However, further research would be necessary to validate these test based measures as a proxy for non-cognitive skills.

We find similar results when using two questions of the background questionnaire that are related to individuals' personality to construct self-reported measures of noncognitive skills: political efficacy and social trust. However, the same is not true for cultural engagement.

In addition, we use our constructs to provide corrected measures of PIAAC scores that try to net-out the effect of non-cognitive skills. By doing so, we observe that the relationship between earnings and PIAAC scores might be partially mediated by non-cognitive skills. However, our results also show that PIAAC test scores remain highly significant and as the main determinant of earnings after the correction. Given that our strategy might be seen as an over correction of PIAAC scores, this result highlights the prominent role of cognitive skills in the earnings equation and reinforces the results of Hanushek et al. (2015).

Finally, our study reveals the necessity of including items in the international assessments that facilitate information and measurement of non-cognitive skills that can complement the already existing information on cognitive skills.

Open Access This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made.

References

Allport GW (1961) Pattern and growth in personality. Holt, Rinehart and Winston, New York

- Almlund M, Duckworth AL, Heckman J, Kautz T (2011) Personality psychology and economics. In: Hanushek EA, Machin S, Woessmann L (eds) Handbook of the economics of education, vol 4. North Holland, Amsterdam
- Almond GA, Verba S (1965) The civic culture: political attitudes and democracy in five nations. Princeton university press, Princeton
- Balart P (2017) Parental non-response and students' test scores. Learn Individ Differ 55:141-149
- Becker GS (1962) Investment in human capital: a theoretical analysis. J Polit Econ 70:9–49
- Boe EE, May H, Boruch R (2002) Student task persistence in the third international mathematics and science study: a major source of achievement differences at the national, classroom, and student levels. Center for research and evaluation in social policy. CRESP-RR-2002-TIMSS1
- Borghans L, Schils T (2012) The leaning tower of Pisa: decomposing achievement test scores into cognitive and noncognitive components. (unpublished manuscript)
- Borghans L, Meijers H, Ter Weel B (2008) The role of noncognitive skills in explaining cognitive test scores. Econ Inq 46(1):2–12
- Borghans L, Golsteyn BH, Heckman JJ, Humphries JE (2011) Identification problems in personality psychology. Personal Individ Differ 51:315–320
- Brunello G, Schlotter M, (2011) Non-cognitive skills and personality traits: labour market relevance and their development in education and training systems. IZA Discussion Paper No. 5743
- Cabrales A, Dolado JJ, Mora R (2014) Dual labour markets and (lack of) on-the-job training: PIAAC evidence from Spain and other EU countries. IZA Discussion Paper No. 8649
- Campbell A (1980) The American voter. University of Chicago Press, Chicago

Campbell A, Gurin G, Miller WE (1954) The voter decides. Row, Peterson, Evanston

Card D (1999) The causal effect of education on earnings. In: Ashenfelter O, Card D (eds) Handbook of labor economics, vol 3a. North Holland, Amsterdam, pp 1801–1863

- Cheng, A, Zamarro G, Orriens B (2016) Personality as a predictor of unit nonresponse in panel data: an analysis of an internet-based survey. CESR-Schaeffer working paper No. 2016-014
- Cunha F, Heckman JJ (2008) Formulating, identifying and estimating the technology of cognitive and non-cognitive skill formation. J Hum Resour 43(4):738–782
- Cunha F, Heckman JJ, Schennach SM (2010) Estimating the technology of cognitive and non-cognitive skill formation. Econometrica 78:883–931
- Duckworth AL, Quinn PD, Lynam DR, Loeber R, Stouthamer-Loeber M (2011) Role of test motivation in intelligence testing. Proc Natl Acad Sci 108:7716–7720
- Efron B (1982) The jackknife, the bootstrap and other resampling plans. SIAM, Philadelphia
- García E (2016) The need to address non-cognitive skills in the education policy agenda. In: Khine MS, Areepattamannil S (eds) Non-cognitive skills and factors in educational attainment. Contemporary approaches to research in learning innovations. Sense Publishers, Rotterdam
- Gneezy U, Rustichini A (2000) Pay enough or don't pay at all. Q J Econ 115(3):791-810
- Grönqvist E, Öckert B, Vlachos J (2017) The intergenerational transmission of cognitive and non-cognitive abilities. J Hum Resour 52(4):887–918
- Grotlüschen A (2017) The narrative of 'equality of chances' as an approach to interpreting PIAAC results on perceived political efficacy, social trust and volunteering and the quest for political literacy. Eur J Res Educ Learn Adults 8(2):1–17
- Groves RM, Fowler FJ Jr, Couper MP, Lepkowski JM, Singer E, Tourangeau R (2011) Survey methodology. Wiley, New York
- Hanushek EA, Woessmann L (2008) The role of cognitive skills in economic development. J Econ Lit 46(3):607–668
- Hanushek EA, Woessmann L (2011) The economics of international differences in educational achievement. In: Hanushek EA, Machin S, Woessmann L (eds) Handbook of the economics of education, vol 3. North Holland, Amsterdam
- Hanushek EA, Schwerdt G, Wiederhold S, Woessmann L (2015) Returns to skills around the world: evidence from PIAAC. Eur Econ Rev 73:103–130
- Heckman JJ, Polachek S (1974) Empirical evidence on the functional form of the earnings–schooling relationship. J Am Stat Assoc 69(346):350–354
- Heckman JJ, Rubinstein Y (2001) The importance of non-cognitive skills: lessons from the GED testing program. Am Econ Rev 91:145–149
- Heckman JJ, Stixrud J, Urzua S (2006) The effects of cognitive and non-cognitive abilities on labor market outcomes and social behavior. J Labor Econ 24(3):411–482
- Heckman JJ, Cunha F, Schennach S (2010) Estimating the technology of cognitive and noncognitive skill formation. Econometrica 78(3):883–931
- Hedengren D, Stratmann T (2012) The dog that didn't bark: what item nonresponse shows about cognitive and non-cognitive ability. SSRN working paper No. 2194373
- Hernández M, Hershaff J (2014) Skipping questions in school exams: the role of socio-emotional skills on educational outcomes. Mimeo, New York
- Hitt C, Trivitt J, Cheng A (2016) When you say nothing at all: the predictive power of student effort on surveys. Econ Educ Rev 52:105–119
- Kaiser HF (1960) The application of electronic computers to factor analysis. Educ Psychol Meas 20:141-51
- Kautz T, Heckman JJ, Diris R, Ter Weel B, Borghans L (2014) Fostering and measuring skills: improving cognitive and non-cognitive skills to promote lifetime success. Technical report, National Bureau of Economic Research
- King G, Murray C, Salomon J, Tandon A (2004) Enhancing the validity and crosscultural comparability of measurement in survey research. Am Polit Sci Rev 98:567–583
- Levy PS, Lemeshow S (1999) Sampling of populations: methods and applications, 3rd edn. Wiley, New York
- McCrae RR, John OP (1992) An introduction to the five-factor model and its applications. J Personal 60(2):175–215
- Mincer J (1970) The distribution of labor incomes: a survey with special reference to the human capital approach. J Econ Lit 8:1–26
- Mincer J (1974) Schooling, experience, and earnings, human behavior and social institutions, No. 2
- OECD (2013) OECD first results from the survey of adult skills. OECD Publishing, Paris
- OECD (2016) Technical report of the survey of adult skills (PIAAC), 2nd edn. OECD Publishing, Paris Paulhus DL (1984) Two-component models of socially desirable responding. J Personal Soc Psychol 46:598

- Riphahn RT, Serfling O (2005) Item non-response on income and wealth questions. Empir Econ 30:521-538
- Segal C (2012) Working when no one is watching: motivation, test scores, and economic success. Manag Sci 58:1438–1457
- Shoemaker PJ, Eichholz M, Skewes EA (2002) Item nonresponse: distinguishing between don't know and refuse. Int J Public Opin Res 14:193–201
- Uslaner EM (1999) Democracy and social capital. In: Warren M (ed) Democracy and trust. Cambridge University Press, Cambridge, pp 121–50
- Uslaner EM (2000) Producing and consuming trust. Polit Sci Q 115(4):569-590
- Van Soest A, Delaney L, Harmon C, Kapteyn A, Smith JP (2011) Validating the use of anchoring vignettes for the correction of response scale differences in subjctive questions. J R Stat Soc Ser A Stat Soc 174(3):575–595
- Wechsler D (1940) Nonintellective factors in general intelligence. J Abnorm Soc Psychol 38(1):101
- Welland JD (1978) Cognitive abilities, schooling and earnings: the question of functional form. Rev Econ Stat 60(4):622–627
- Zamarro G, Cheng A, Shakeel D, Hitt C (2016a) Comparing and validation measures of character skills: finding from a nationally representative sample. EDRE working paper 2016-08
- Zamarro G, Hitt C, Mendez I (2016b) When students don't care: reexamining international differences in achievement and non-cognitive skills. EDRE working paper No. 2016-18
- Zamarro G, Nichols M, Duckworth A, D'Mello S (2017) Further validation of survey-effort measures of conscientiousness: results from a sample of high school students. Mimeo, New York

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.